Stock Market Liberalizations and Export Dynamics^{*}

Melise $JAUD^{\dagger}$ Madina KUKENOVA[‡] Martin STRIEBORNY[§]

This version: June 5, 2019

Abstract

We find that granting foreign investors access to domestic stock markets facilitate efficiency-enhancing product specialization in the recipient countries. After countries liberalize their stock markets and allow foreign investors to acquire equity stakes in domestic firms, products that do not correspond to the liberalizing countries' comparative advantage disappear disproportionately faster from their export portfolios. At the same time, the overall long-term export performance of the liberalizing countries improves. Domestic stock market development does not play the same disciplining role in forcing termination of inefficient exports, suggesting a unique role for foreign investors in improving resource allocation in the real economy.

JEL classification: G15, F65, O16.

Keywords: stock market liberalizations, disciplining role of foreign investors, export dynamics

^{*}We would like to thank for very useful comments and discussions Sumit Agarwal, Caroline Freund, Björn Hagströmer, Eva Liljeblom, Taylan Mavruk (discussant), Lars Nordén, Lars Oxelheim, Daniel Paravisini, Kelly Shue, Neeltje van Horen, Andrei Zlate (discussant), and the seminar and conference participants at Adam Smith Business School at University of Glasgow, Toulouse Business School, Aarhus University, Stockholm Business School, University of Groningen, IESEG School of Management, Trinity Business School, Lund University, Financial Management Association Annual Meeting in San Diego, Econometric Society European Summer Meeting in Cologne, Econometric Society North American Summer Meeting in Davis, Gothenburg-Lund Workshop in Finance, Zeuthen Workshop on International Financial and Monetary Architecture, and Conference on Trade and Integration in a Time of Anti-Globalization. The paper is part of a World Bank program. The authors appreciate the financial support from Webster University Geneva. Melise Jaud appreciates the financial support from the Swiss National Science Foundation. Martin Strieborny appreciates the financial support from the Handelsbanken Research Foundation. The most recent version of this paper including an accompanying Online Appendix can be downloaded from www.martinstrieborny.com

[†]World Bank; E-Mail: mjaud@worldbank.org

[‡]Webster University Geneva; E-Mail: kukenova@webster.ch

[§]Knut Wicksell Centre for Financial Studies, Lund University, E-Mail: Martin.Strieborny@nek.lu.se

1 Introduction

Is financial openness beneficial for the real economy? What is the economic impact of foreign investors in the recipient countries? Answering such questions is as relevant as ever, given the current policy climate characterized by growing "economic nationalism" and increasing resistance against the current levels of financial and economic integration. Existing literature on the topic focuses primarily on aggregate outcomes, and tends to find a positive effect of cross-border financial integration on domestic investment (Henry 2000a), economic growth (Bekaert, Harvey, and Lundblad 2005; Gupta and Yuan 2009), and international trade (Manova 2008) while also raising concerns about potential adverse effects in terms of more volatile capital flows (Broner and Ventura 2016) and increased vulnerability to financial crises (Reinhart and Rogoff 2009). Less attention has been devoted to the potential structural changes brought by foreign investors to recipient economies (Bekaert et al. 2007; Larrain and Stumpner 2017).

This paper asks whether foreign investors can trigger a positive structural change in the real economy by identifying and promoting the most "promising" industries and products in the recipient countries. A positive answer to this question would be in accordance with two influential strands of finance literature arguing that financial development and financial openness improve allocation of available resources (e.g., Wurgler 2000; Fisman and Love 2004; Bekaert et al. 2007; Larrain and Stumpner 2017) and that foreign investors can increase efficiency through their superior monitoring skills (e.g., Gillan and Starks 2003; Ferreira and Matos 2008; Aggarwal et al. 2011; Bena et al. 2017; Luong et al. 2017).

To explore the structural change brought by foreign investors, we utilize one of the major economic insights stating that countries should specialize in exporting products where they possess comparative advantage. In particular, we examine whether the entry of foreign investors facilitates efficiency-enhancing specialization by better aligning export portfolio of a given country with its comparative advantage. Framing the research question within the context of international trade allows us to perform our empirical analysis with the help of highly disaggregated data at the product level that are both internationally comparable and available for a large set of different countries.

The use of product-level data in our analysis helps us to sidestep the issue of spillovers and economic externalities that could arise in the context of firm-level data. Focusing the empirical analysis only on firms acquired by foreign investors might namely fail to capture the overall structural impact of financial liberalization. For instance, foreign-owned firms can generate substantial spillovers in terms of productivity and corporate governance both to their competitors (e.g., Albuquerque et al. 2019) and to their suppliers (e.g., Javorcik 2004).¹ Opening up of domestic economy to foreign investors can also generate across-theboard spillovers at the country level, either by providing better incentives for domestic corporate insiders to improve corporate governance (Doidge, Karolyi, and Stulz 2007) or by triggering governmental reforms of the public governance and corporate regulation (e.g., Stulz 2005; Kaminsky and Schmukler 2008). The potential bias in estimating the structural impact of foreign investors by comparing the performance of acquired and non-acquired domestic firms could be further exacerbated if some of these spillovers substantially elevate the stock prices of the remaining domestic firms, making them less likely to be ever acquired by foreign investors. This so-called "feedback effect" has been so far analyzed mostly in the domestic context of US firms and investors (see, e.g., Bradley et al. 2010; Edmans, Goldstein, and Jiang 2012), but it could also affect an empirical analysis focusing on the structural impact of foreign investors.

Our empirical strategy consists in examining disaggregated export flows from multiple countries that at different points of time allow foreign investors to acquire equity stakes in domestic firms. We rely on UN Comtrade data that comprise detailed information on worldwide trade flows for almost 1,300 different product categories. We combine these data with the stock market liberalization dates from Bekaert, Harvey, and Lundblad (2005) that capture the moments in time when governments of different countries allow for the first time foreign investors to acquire equity stakes in domestic firms. These stock market liberalization events thus allow for the estimation of a staggered treatment of financial integration and are often used as quasi-natural experiments to explore the impact of foreign investors on domestic economy (see the review by Henry 2006 as well as Bekaert and Harvey 2000; Henry 2000a, 2000b; Bekaert, Harvey, and Lundblad 2005; Manova 2008; Gupta and Yuan 2009; Defever and Suedekum 2014).

Using the combined product-country-time data, we explore in a difference-in-difference framework whether stock market liberalization events push countries' export portfolios closer to their comparative advantage. We do find that this is indeed the case. After

¹ In the domestic US context, Aslan and Kumar (2016) and Gantchev, Gredil, and Jotikasthira (2018) explore spillovers from firms targeted by the hedge fund activism to the non-targeted firms. The transmission channels they identify (product market competition, attempts by the management of the non-targeted firms to avoid hostile activism themselves) might also be relevant in the case of foreign investors.

foreign investors are allowed to invest in domestic stock market, the products that do not correspond to the comparative advantage of the liberalizing country face significantly worse chances of remaining in the country's export portfolio compared to products that are well aligned with the comparative advantage. At the same time, the entry of foreign investors into domestic stock market improves the overall long-term export performance of the liberalizing country in the world market. These results are robust to controlling for various other financial and non-financial channels, including the strength of domestic banks and stock markets, economic development, trade openness, and trade liberalization events.

This paper is related to three strands of literature. First, we contribute to the literature examining the impact on the real economy of international financial integration in general and stock market liberalization in particular. Most of the papers in this area have investigated the impact of financial liberalization on investment in physical capital, economic growth or production volatility (see, e.g., Henry 2000a; Bekaert, Harvey, and Lundblad 2005; Gupta and Yuan 2009; Levchenko, Ranciere, and Thoenig 2009).² Papers that analyze the impact of stock market liberalization on international trade within a country-industry setting include Manova (2008) and Defever and Suedekum (2014). In our paper, we use the framework of international trade to analyze the impact of foreign investors on structural change and resource allocation at the product level. The empirical research on allocative consequences of financial liberalization usually faces a trade-off between the level of disaggregation and the cross-country coverage and comparability of the available data. Researchers thus often have to choose between maximizing the number of countries across different stages of financial and economic development (e.g., Bekaert et al. 2007) and utilizing firm-level data (e.g., Larrain and Stumpner 2017).^{3,4} Our use of international trade data allows us to explore the allocative consequences of financial liberalization in the context of 81 countries exporting 1,299 products. Moreover, the

² Papers examining the impact of financial liberalization on the outcomes in financial markets rather than in real economy include, e.g., Bekaert and Harvey (2000), Henry (2000b), Kaminsky and Schmukler (2008).

³ Bekaert et al. (2007) use a panel data of 50 countries between 1980 and 2002 and show that financial openness is an important factor in aligning growth opportunities with actual growth. Larrain and Stumpner (2017) use firm-level data from 10 post-communist countries in Central and Eastern Europe and provide evidence that capital account liberalization helps financially constrained firms to demand more capital and produce at more efficient levels. Also related is the work by Varela (2018) that looks at impact of lifting the restrictions on international borrowing in the case of Hungarian firms.

⁴ Related research examining the impact of domestic financial development rather than of foreign investors on allocation of physical capital includes, e.g., Wurgler (2000), Beck and Levine (2002), Love (2003), and Fisman and Love (2004, 2007).

framework of international trade allows us to move beyond the prevailing research focus on the misallocation of physical capital and examine the allocative impact of financial liberalization also under consideration of human capital and natural resources.⁵

Second, we contribute to the literature examining the disciplining impact of foreign investors. Several influential papers argue that foreign investors are more effective as external monitors because they lack business relations with local firms that often induce domestic investors to feel loyal to the local management (e.g., Gillan and Starks 2003; Ferreira and Matos 2008; Aggarwal et al. 2011; Bena et al. 2017; Luong et al. 2017).⁶ Our results are consistent with this argument. On one hand, we show that allowing foreign investors to acquire equity in domestic firms pushes the export portfolio of the liberalized economy closer towards its comparative advantage. On other hand, we do not find any disciplining impact of domestic stock market development when it comes to termination of inefficient exports, with this latter result being also in accordance with previous research (Jaud, Kukenova, and Strieborny 2018). Our findings thus seem to support the strand of literature stressing the unique ability of foreign investors to increase efficiency through their superior monitoring abilities. Section 6 provides further corroborative evidence in this direction.

Third, the paper is related to the literature examining the determinants of establishing a long-term successful presence in the foreign markets. Besedes and Prusa (2006a) were the first to apply the formal survival analysis in the context of international trade and showed how short-lived most of the exports are. Subsequent research confirmed that most new exports vanish within the first year or two of their existence. Long-term export survival thus seems a particularly important export margin that often separates successful countries from the unsuccessful ones (see, e.g., Besedes and Prusa 2011). And while there has been research on the impact of domestic financial development on long-term export survival (Jaud, Kukenova, and Strieborny 2015, 2018), we are not aware of a systematic

⁵ In particular, we base our empirical analysis on the Heckscher-Ohlin theory of international trade maintaining that exported products should intensively use those factors of production that are abundant and therefore cheap in the exporting country. The analyzed factors of productions include physical capital, human capital, and natural resources. Subsection 2.2 provides more details.

⁶ Recent research about finance and relationship-specific investment (Strieborny and Kukenova 2016; Strieborny 2017) also indicates a special role for the foreign investors that goes beyond the development of domestic stock markets in general.

analysis of the role of financial liberalization in this regard.⁷

The rest of the paper is organized as follows. Section 2 describes the data, and Section 3 outlines our estimation strategy. Section 4 presents the main empirical results, and Section 5 reports the results of robustness checks. Section 6 provides some evidence that the economic mechanism driving our results relates to the monitoring by foreign investors. Section 7 concludes and suggests directions for further research.

2 Data

Our unit of observation is an export spell - a continuous exporting of product k from country c to the world market. The final sample contains data on 1,299 products exported from 81 countries and focuses on export spells that started during the period of extensive financial liberalization between 1980 and 1997. The term "time" thus refers in this paper to the year when an export spell started unless stated otherwise. We follow these spells until 2006, the year before the global financial crisis started. This approach allows us to examine the impact of stock market liberalization events in 1980s and 1990s on the long-term export survival, in accordance with our focus on the structural change in the real economy brought about by the presence of foreign investors.

The period of 1980s and 1990s represents a fitting environment for examining the impact of financial liberalization on the real economy. Out of the eighty-one exporting countries in our sample, thirty-six countries experienced stock market liberalization between 1980 and 1997. A significant number of financially closed countries thus opened up their domestic stock markets to foreign investors during this period, with different countries allowing foreign investors to acquire equity stakes in domestic firms in different years. At the same time, this wave of staggered financial liberalization happened before the across-the-board explosion of cross-border financial flows in 2000s that preceded the global financial crisis.⁸

⁷ There has also been an extensive research on both trade survival in general (e.g., Besedes and Prusa 2006b; Nitsch 2009; Brenton, Saborowski, and von Uexkull 2010; Cadot et al. 2013) and on impact of finance on other margins of international trade like export entry or export volume (e.g., Beck 2002, 2003; Amiti and Weinstein 2011; Becker, Chen, and Greenberg 2013; Manova 2013; Paravisini et al. 2015).

⁸ These later cross-border financial flows could also affect the long-term survival of the export spells that started during the 1980-1997 period. Due to their across-the-board character, they would probably introduce bias against finding any differential impact of the previous staggered stock market liberalization events. Nevertheless, in one of the robustness tests reported in Table 9, we stopped following the export spells already in the year 1997 rather than 2006.

2.1 Financial variables

A stock market liberalization event occurs when government of a country opens up its domestic stock market to the presence of foreign investors, allowing them to acquire equity stakes in domestic companies. Our data on stock market liberalization come from Bekaert, Harvey, and Lundblad (2005) that cover the period of extensive financial liberalization between 1980 and 1997.⁹ Bekaert, Harvey, and Lundblad (2005) report the official year of the stock market liberalization, the year of a first sign of stock market liberalization, and a continuous variable capturing the liberalization intensity that represents the proportion in the portfolio of domestic equities the foreign investors can acquire. We focus on the official stock market liberalization dates in the main text and perform a series of robustness tests using the dummy for the first sign of liberalization and a continuous variable capturing the intensity of stock market liberalizations in the Online Appendix D. As discussed in more detail at the end of Section 5, the results reported in Online Appendix D are generally stronger than the ones reported in the main text as the use of official stock market liberalization dates introduces bias against finding any significant results.

Based on the liberalization dates from Bekaert, Harvey, and Lundblad (2005), we construct a liberalization dummy that is equal to one if a given country allows in a given year foreign investors to acquire shares in the domestic firms and zero otherwise. The liberalization dummy thus always equals one for the countries that implemented stock market liberalization prior to 1980. By the same token, the dummy always equals zero for the countries that did not liberalize their stock market until 1997. For countries that opened up their stock market to foreign investors during the period 1980-1997, the liberalization dummy takes value one if a given export spell is exposed to the liberalization event, i.e. if the official stock market liberalization date occurs before or during the life of the spell.

As for other financial variables, our proxies for the level of domestic banking and stock market development in the liberalizing countries are the ratio of private credit to GDP and the ratio of stock market capitalization to GDP, respectively. Both variables are

⁹ Appendix A provides a list of the countries. The original database of Bekaert, Harvey, and Lundblad (2005) covers 91 countries, including 41 countries that liberalized their stock market at some point during the 1980-1997 period, 16 countries that liberalized their stock market prior to 1980, and 34 countries that kept their stock market closed to foreign investors during the whole period under consideration. In our final sample, the number of all countries from 91 to 81 and the number of countries that liberalized their stock market during the period declines from 41 to 36 due to the lack of data for some of the control variables.

from the database by Beck, Demirguc-Kunt, and Levine (2000). The industry measure of dependence on external finance at the ISIC 3-digit level is from Braun (2003). This measure goes back to the seminal paper by Rajan and Zingales (1998) and captures the share of capital expenditures that cannot be financed by internally generated funds. It is computed as ratio of the capital expenditures minus cash flow from operations to the capital expenditures for the median publicly listed company in a given US industry, using the financial data on US firms from Compustat.¹⁰

2.2 Distance to comparative advantage

Our measure of distance to comparative advantage captures how far is a given product from the comparative advantage of a given country, based on the Heckscher-Ohlin theory of international trade. This workhorse model of comparative advantage maintains that exported products should intensively use those factors of production that are abundant and therefore cheap in the exporting country. In other words, the factor intensity of the exported products should correspond to the factor endowment of the country that exports them. Building upon on the notion of "product chain of comparative advantage" within the Heckscher-Ohlin theoretical framework (Jones 1956-57; Bhagwati 1972; Deardorff 1979), we measure the distance to comparative advantage using Euclidean metrics of distance between the vector of the country's factor endowments and the vector of the product's revealed factor intensities:

$$distance_{ck} = \sqrt{std(\kappa_c - \hat{\kappa}_k)^2 + std(h_c - \hat{h}_k)^2 + std(l_c - \hat{l}_k)^2},\tag{1}$$

where κ_c , h_c and l_c are the country's endowments in physical capital, human capital and land, and $\hat{\kappa}_k$, \hat{h}_k and \hat{l}_k are the corresponding revealed factor intensities of product k. We standardize the differences between the product's factor intensities and the country's factor endowments to have zero mean and unit variance ("std" in Equation 1 stands for "standardized"). This allows us to give equal weights to all three production factors, which are measured in different units. The resulting metrics $distance_{ck}$ is time-varying as countries' factor endowments evolve over time, with the time subscript in Equation 1

¹⁰ The idea of using data on publicly listed US companies to compute a generic industrial measure of dependence on external finance relies on the argument that these companies face relatively frictionless financial markets. Consequently, the need for external finance of a median US public company captures the true technologically determined need for external finance at the industry level that is unaffected by various financial and contractual frictions facing smaller firms or firms in less financially developed countries. See Rajan and Zingales (1998) and Braun (2003) for a more detailed discussion.

omitted for simplicity.¹¹

The data on countries' factor endowments and products' revealed factor intensities is taken from Cadot, Tumurchudur, and Shirotori (2009). Physical capital per worker is constructed according to the perpetual inventory method. Human capital endowment is proxied by the average years of schooling. The land endowment corresponds to the number of hectars of arable land per person in a given country. The factor intensities for a product k are calculated as the weighted average of the factor endowments of the countries exporting this product. The revealed physical capital intensity of good k is calculated as:

$$\hat{\kappa}_{k,t} = \sum\nolimits_{c} \omega_{ck,t} \kappa_{c,t},$$

where κ_c is country c's endowment in physical capital at time t, and the weights are given by $\omega_{ck,t} = \frac{X_{ck,t}/X_{c,t}}{\sum_c X_{ck,t}/X_{c,t}}$ with X_c representing value of all products exported from country c and X_{ck} value of exports of product k from country c. The weight $\omega_{ck,t}$ will be thus higher for countries with a higher $X_{ck,t}/X_{c,t}$ ratio, i.e. for countries where product k represents a higher share in the country's overall exports. Consequently, the more a given country specializes in exporting a given product, the higher influence has the factor endowment of that country on that product's revealed factor intensity. The revealed human capital intensity and land intensity of product k are calculated in a similar way.¹²

2.3 Other variables

We use export data from the UN Comtrade database. The data is reported at the level of 4 or 5 digits of the SITC product classification and comprises 1,299 different product categories. We construct a survival database focusing on the information about the starting and ending year of a trade relationship. To correct for left-censored observations (i.e., export spells that are already in place at the beginning of the sample) and to match the survival database with the period covering the stock market liberalization events, we drop all export spells that started before 1980 and after 1997.

¹¹ The specific metrics of distance to comparative advantage used here is a slightly modified version of a measure introduced by Cadot, Carrere, and Strauss-Kahn (2011). See Jaud, Kukenova, and Strieborny (2018) for a more detailed discussion.

¹² The original index of revealed comparative advantage goes back to the seminal paper by Balassa (1965). The approach used by Cadot, Tumurchudur, and Shirotori (2009) builds upon the methodology introduced by Hausmann, Hwang, and Rodrik (2007). Their formulation applies the normalization in the denominator of $\omega_{ck,t}$ to ensure that the weights add up to one when aggregating across all countries that export a given product.

We construct several spell characteristics using the original export data. We compute the number of suppliers (i.e., exporting countries) of a given product to the world market (measured at the product-time level), initial export value of a given product (countryproduct-time level), total value of country's exports to the rest of the world (country-time level), number of export spells during the sample period for a given country-product pair (country-product level). These variables account for the level of competition and/or the potential market size for a given product, the degree of confidence between partners at the initiation of a trade relationship, the size and overall export performance of the country, and the possibility that repeated exits and re-entries into exporting affect the chances for the long-term export survival.

The physical and human capital intensities at the level of ISIC 3-digits industrial sectors are taken from Braun (2003). The data identifying products as homogeneous, differentiated or price-referenced in trade publications (an intermediate category) is from Rauch (1999). The annual data for GDP per capita in the purchasing power parity terms (GDP pc in PPP terms) is taken from the World Development Indicator Report 2006, and is reported in constant 2000 US dollars. The data for real exchange rate is also taken from the World Development Indicators database. The trade openness variable (the sum of country's exports and imports divided by its GDP) is from the Penn World Tables. The IMF dummy capturing whether a given country was a subject to the IMF program (i.e., received an official IMF loan) in a given year is constructed based on data from the International Monetary Fund.

Trade liberalization dates come from the database of Wacziarg and Welch (2008) who extended and refined the trade polices database of Sachs and Warner (1995). Sachs and Warner (1995) define trade openness based on five conditions that capture trade policy. A country is open to trade if it meets *none* of the following criteria: Average tariff rates of 40 percent or more; Nontariff barriers covering 40 percent or more of trade; A black market exchange rate at least 20 percent lower than the official exchange rate; A state monopoly on major exports; A socialist economic system. In principle, Sachs-Warner's trade liberalization date is the date after which all of the Sachs-Warner openness criteria are continuously satisfied. However, due to lack of available data, not all trade liberalization dates identified by Sachs and Warner satisfy the five criteria of trade openness. Wacziarg and Welch (2008) updated trade liberalization dates based on new information which became available after the study of Sachs and Warner (1995). In addition to primary-source data on annual tariffs, nontariff barriers, and black market

premium used by Sachs and Warner (1995), Wacziarg and Welch (2008) rely on a variety of secondary sources to identify the dates of abolishment of export marketing boards and the dates when multiparty governance systems replaced communist regimes.

The data on Legal System and Property Rights used in Section 6 come from the Economic Freedom Database by Fraser Institute. The index captures the degree of protection of property rights in the countries for the year 1980. The index varies from 0 to 10, with the lower number corresponding to lower level of investor protection. Gwartney, Lawson, and Norton (2008) provide more details. The alternative measures of legal environment used in Online Appendix E (Efficiency of Judicial System, Rule of Law, Risk of Expropriation, Repudiation of Contracts by Government, and Investor Protection - an average of the three previous measures) are based on La Porta et al. (1998). The original sources of data for these alternative measures are from Business International Corporation and International Country Risk Guide (see La Porta et al. 1998 for details).

2.4 Preliminary evidence and descriptive statistics

This paper explores whether allowing foreign investors to acquire equity stakes in domestic firms triggers an efficiency-enhancing structural change, pushing the country's exporters to re-balance their export portfolios towards products consistent with the country's comparative advantage. Figure 1 and Figure 2 provide some preliminary graphical evidence supporting this hypothesis in the context of the long-term export survival. The two figures show Kaplan-Meier survival functions capturing the probability of a new export spell (continuous exporting of product k from country c to the world market) to survive after year 1,2, etc. Given the annual frequency of the data, the survival probability in the first year is one by default.

Figure 1 and Figure 2 show the average export survival rate for two categories of products - products that are in the 25th and 75th percentile of distance to comparative advantage, i.e. products that are close versus far away from the comparative advantage of the exporting country. Only countries that did liberalize during the period 1980-1997 are included. For these countries, Figure 1 reports the survival rate before and Figure 2 the survival rate after the event of equity market liberalization. The comparison of the two figures reveals that the overall export performance of countries improves after they allow foreign investors to acquire equity stakes in domestic companies - both survival functions move upwards. However, this effect is not uniform across products. The exported products that do correspond to the comparative advantage of the liberalizing countries

benefit disproportionately more from the entry of foreign investors. The survival function for the products in the 25th percentile of distance to comparative advantage (blue line) moves upward more strongly and is much clearer above the survival function of products in the 75th percentile of distance to comparative advantage (red line) in Figure 2 compared to Figure $1.^{13}$

Table 1 reports descriptive statistics regarding the export spells in our sample that supports the preliminary evidence from Figure 1 and Figure 2. The first two rows of Table 1 show that both average and median duration of export spells increases after a financially closed country allows foreign investors to acquire equity stakes in domestic firms. The third and fourth row of Table 1 capture the situation in the liberalizing subsample before the event of stock market liberalization and can be thus seen as an analogue to Figure 1. We can see that the export performance of products that correspond to comparative advantage of a given country (bottom 25th percentile of distance to comparative advantage - DCA) is practically indistinguishable from the export performance of products that are far away from the comparative advantage (top 25th percentile of DCA). The fifth and sixth row (an analogue to Figure 2) reflect the situation after a country grants foreign investors access to its domestic stock market. The overall length of export spells increases for both categories of products, but the products corresponding to the comparative advantage benefit disproportionately more (average length increasing to 7.78 years and median length to 4 years) compared to products that are far away from the comparative advantage of the liberalizing country (average length increasing to 6.46 years and median length to 3 years). The bottom part of Table 1 repeats the whole exercise for the full sample of countries including also countries whose stock markets remained continuously closed or continuously open to foreign investors during the whole 1980-1997 period.

The preliminary evidence in Figure 1, Figure 2, and Table 1 suggests that allowing foreign investors to acquire equity stakes in domestic firms indeed helps countries to rebalance their export portfolios towards their comparative advantage, making a better use of available resources. However, this evidence does not control for various alternative channels and other confounding factors at country, industry, and product level. Before introducing our formal regression framework in the next section, Table 2 reports the

¹³ Note that the percentiles are defined on a country-by-country basis. Consequently, the graphs capturing the differential survival of products that are close versus far away from comparative advantage of liberalizing countries are not biased by differences across individual countries' alignment of their overall export portfolio with their comparative advantage.

summary statistics for the variables included in the later empirical analysis.

3 Estimation Strategy

Our empirical approach relies on the use of difference-in-difference estimation strategy within the econometric framework of the survival (also called duration) analysis. The difference-in-difference approach allows us to examine whether the entry of foreign investors affects exports differently depending on the alignment of individual products with the comparative advantage of a given country. The survival analysis enables us to focus on the long-term structural impact of foreign investors in this regard, by looking at the export survival of individual products as our outcome variable.

We analyze the long-term export survival by estimating the widely used Cox Proportional Hazard Model (Cox 1972) that is described in more detail in Appendix B1. The specification examining the impact on export survival of allowing foreign investors to invest in domestic stock market writes:

$$h_{ck}(t) = h_{0,k}(t) \exp[\beta_1 StMLib_{c,t_0} + \beta_2 StMLib_{c,t_0} * distance_{ck,t_0} + \beta_3 * distance_{ck,t_0} + \mathbf{Y}_{ck,t_0}\phi + \delta_c + \delta_{t_0} + \varepsilon_{ck,t_0}]$$
(2)

where $StMLib_{c,t_0}$ is the stock market liberalization dummy, $distance_{ck,t_0}$ is a distance of exported product k from the comparative advantage of exporting country c, δ_c are exporting-country fixed effects, δ_{t_0} are time fixed effects (with time referring to the year when the export spell started), and ε_{cikj} is a stochastic error term. All explanatory variables are measured at the beginning of the export spell (at time t_0), except for the stock market liberalization dummy, which takes value 1 if the liberalization event occurs before the initiation of a spell or during the life of a spell.¹⁴ Since the stock market liberalization variable varies across time and countries, we cluster the reported standard errors at the country-time level.

The dependent variable is the hazard rate of an export spell of product k from country c to the world market. The hazard rate captures the probability of export failure rather than export success and can be thus seen as a mirror image of the Kaplan-Meier survival functions reported in Figure 1 and Figure 2. In the Cox Proportional Hazard Model, the

 $^{^{14}}$ We slightly abuse the notation and use subscript t_0 also in the case of stock market liberalization: $StMLib_{c,t_0}$

outcome variable is thus the probability of an export spell that started at time t_0 to fail at time t.

Our main variable of interest is the interaction between stock market liberalization and the product's distance to comparative advantage while we also allow the stock market liberalization and the distance to comparative advantage to enter the regression directly. The main interaction term $StMLib_{c,t_0}*distance_{ck,t_0}$ tests whether stock market liberalization has a differentiated effect on survival of different product categories. In particular, a positive coefficient β_2 would suggest that products relying intensively on scarce factors disappear at a faster rate from country's export portfolio after foreign investors are allowed to acquire equity stakes in domestic companies.

 Y_{ck,t_0} is the vector of control variables. Depending on specification, it contains various interaction terms controlling for alternative channels that could be correlated with the mechanism captured by the main interaction term $StMLib_{c,t_0} * distance_{ck,t_0}$. In particular, we interact the stock market liberalization with the dependence on external finance at the industry level to control for the possibility that stock market liberalizations disproportionately benefit industries that require a high level of external finance. Furthermore, we control for the possible differential impact of domestic banks and stock markets by interacting distance to comparative advantage both with the ratio of stock market capitalization over GDP and with the ratio of bank credit to private sector over GDP. To control for non-financial alternative channels, we also include into several specifications the interactions of distance to comparative advantage with GDP per capita in the PPP terms, with trade openness, and with the trade liberalization dummy. Furthermore, we include in almost all specifications (except for the first two baseline ones) the interaction terms of physical and human capital endowment at the country level with the intensity of usage of physical and human capital at the industry level.

The product strata effects discussed in the last paragraph of this section absorb the direct effects of industry characteristics like dependence on external finance or intensity of usage of physical and human capital. By contrast, the control variables at the level of exporting country are time-varying and therefore *not* absorbed by the exporting-country fixed effects. The time-varying country-level variables include stock market development, bank development, GDP per capita in the PPP terms, trade openness, trade liberalization dummy, physical capital per worker, human capital per worker (average years of schooling), and land endowment (arable land per worker).

Finally, we control in all regressions for various product-level variables that could affect

the long-term export survival in the world market. We control for the number of active suppliers (i.e., exporting countries) of a product to the world market that represents the level of competition but also the potential market size for a given product. We also control for the logarithm of export value at the beginning of the spell (initial export), reflecting the degree of confidence between partners at the initiation of the trade relationship. The total value of country's exports to the rest of the world accounts for the size and overall export performance of the country. The number of export spells during the sample period for a given country-product pair captures the possibility that repeated exits and re-entries into exporting of given product affect the chances of the long-term export survival in the world market.¹⁵

We allow the shape of the baseline hazard function $h_{0,k}(t)$ to vary across products, by fitting a stratified Cox Proportional Hazard Model with the SITC product code as a stratification variable. Stratification by products adds more flexibility to the model and can be viewed as a more general way (compared to simple product fixed effects) of accounting for time-invariant unobserved characteristics of the products that affect the probability of export exit. See Appendix B2 for more details.

4 Main Empirical Results

All our estimations investigate exports at the product level from individual countries to the rest of the world. Table 3 reports baseline regressions examining the differential impact of stock market liberalization on the long-term export survival of products, depending on the level of products' congruence with the comparative advantage of the exporting country. Table 4 examines the robustness of our mechanism to the inclusion of alternative channels from finance to real economy. Table 5 focuses on various nonfinancial alternative channels that could also be correlated with our mechanism. Table 6 controls simultaneously for both financial and non-financial channels.

The odd-numbered columns report results for the full sample of countries including those countries where domestic stock market was closed or open to foreign investors during the whole period of 1980-1997. The even-numbered columns report results for the subsample of countries where the event of stock market liberalization occurred between 1980 and 1997. The estimations in the even-numbered columns provide arguably a cleaner

¹⁵ Another standard way to control for the presence of repeated exits and re-entries into exporting within the survival framework consists in introducing a multiple spell dummy that is equal to one if there is more than one spell for a given product-country pair during the sample period. The approach chosen here is more general as it explicitly controls also for the number of these multiple spells.

estimate for our main interaction term capturing the differential impact of the stock market liberalization. The identification comes in this case purely from within-countries changes after the liberalization events. The regression results are thus not potentially biased by structural differences between countries that allowed foreign investors to acquire equity stakes in domestic firms already before 1980 (so that liberalization dummy always equals one for a given country) and countries where stock markets were closed to foreign investors during the whole 1980-1997 period (so that liberalization dummy always equals zero for a given country).

4.1 Baseline Regressions

Table 3 provides first evidence on the impact of stock market liberalization on the longterm survival of exports from liberalizing countries to the rest of the world. All explanatory variables are measured in the year when a given export spell started except for the stock market liberalization dummy that is equal to one if the liberalization date occurred before or during a given export spell, i.e. if an export spell was exposed to the liberalization event. The main variable of interest is the interaction term between the stock market liberalization dummy measured at the country-time level and the distance to the comparative advantage measured at the country-product-time level. Both variables entering the main interaction term (i.e., the stock market liberalization and the distance to comparative advantage) are also directly included in the regressions.

Specifications in columns (1) and (2) of Table 3 include a basic set of control variables: number of suppliers (exporting countries) of a given product to the world market (measured at the product-time level), total value of country's exports to the rest of the world (country-time level), initial export value of a given product (country-product-time level), number of export spells during the sample period for a given country-product pair (country-product level). Specifications in columns (3) and (4) additionally include control variables at the country-time level as well as interactions of those variables with industry characteristics. The direct effects of industry characteristics are captured by the product strata variable. The time-varying country-level control variables include GDP per capita of the exporting country, physical capital per worker, average years of schooling in the country (capturing the level of human capital per worker), and arable land per worker (land endowment). The country-industry interaction terms combine endowments of physical [human] capital at the country level with the intensity of the usage of physical [human] capital at the industry level. These interaction terms control for the fact that exports from industries heavily using physical or human capital might face better survival odds if the exporting country is well-endowed with such capital. Columns (1) and (3) of Table 3 look at the full sample of countries and therefore also include countries where the domestic stock market was continuously open or closed to foreign investors during the whole period of 1980-1997. Columns (2) and (4) include only countries that actually did experience a liberalization event and opened up their domestic stock market to foreign investors between 1980 and 1997.

In all four columns of Table 3, both the stock market liberalization and its interaction with the distance to comparative advantage have the expected sign and are statistically significant. Stock market liberalization decreases the hazard rate of exports and thus improves the survival odds for products exported from the liberalizing country to the rest of the world. However, it helps disproportionately less when it comes to products that do not correspond to the comparative advantage of the exporting country. The positive and significant coefficient for the main interaction term $(StMLib_{c,t_0} * distance_{ck,t_0})$ suggests that the further away is the exported product from the comparative advantage of the exporting country, the higher hazard rate it faces after the stock market liberalization event.

Among the product-level control variables, only initial export and number of spells maintain statistically significant coefficient in all four specifications. The initial export (i.e., the export value at the beginning of a given export spell) that serves as a proxy for the degree of confidence between partners at the beginning of a new trade relationship has the expected negative sign, decreasing the hazard rate of a given export spell. Repeated exits and re-entries into exporting of a given product worsens the long-term export survival (increases the hazard rate) in the world market, as demonstrated by the positive sign for the number of export spells within a given country-product pair during the sample period. Distance to comparative advantage has the expected positive sign, but it is not significant in the last specification reported in column (4). Number of suppliers that can capture both the level of competition and the potential market size also loses significance in the last column of Table 3. The variable total exports controlling for the size of export loses significance in the third column of Table 3.

Similarly to some of the product-level control variables, there is no consistent pattern in terms of significance level also in the case of some of the additional controls at the country and country-industry level that enter the specifications in the last two columns of Table 3. These include GDP per capita, land endowment, and the interaction of physical capital endowment at the country level with the physical capital intensity at the product level. By contrast, the direct effect of physical capital and the effect of the human capital country-industry interaction have a consistent sign and are statistically significant at the 1 per cent level in both column (3) and column (4). The negative coefficients for these two variables suggest that exporting countries' endowment in physical capital decreases the export hazard rate and that human capital endowment disproportionately promotes exports from industries whose production process is particularly intensive in the use of human capital.

Before considering various alternative channels and performing other robustness tests, let us briefly discuss the economic magnitude of the mechanism captured by our main interaction term $(StMLib_{c,t_0} * distance_{ck,t_0})$, with the estimated coefficient β_2 . This term captures the differential impact of equity market liberalization on hazard rate of products that are far away versus products that are close to the comparative advantage of the liberalizing country. For any two products, this differential effect is $\beta_2 * \Delta StMLib_{c,t_0} * \Delta distance_{ck,t_0}$, which in case of liberalization zero-one dummy simplifies to $\beta_2 * (1-0) * \Delta distance_{ck,t_0} = \beta_2 * \Delta distance_{ck,t_0}$. Let us take as an example India, a country that granted foreign investors access to its domestic market in 1992. Five years before the liberalization event, one of the products from the 75th percentile of distance to comparative advantage (DCA) in case of India was "hearing aids" (SITC code 89961, DCA value 1.471), and one of the products from the 25th percentile of distance to Indian comparative advantage was "refined copper included remelted" (SITC code 68212, DCA value 0.482). Using the above formula and taking the estimated coefficient for our main interaction term from column (4) of Table 3 implies that after India liberalized its stock market, the difference between the hazard rates of a product far away from its comparative advantage (hearing aids) relative to hazard rate of product close to its comparative advantage (refined copper) increased by approximately 15 per cent.¹⁶ Appendix C performs this calculation for all countries that have liberalized their stock market between 1980 and 1997 and where data for computing distance to comparative advantage five years before the liberalization event are available. Due to different levels of DCA dispersion within countries' export portfolios, the computed effects capturing the economic magnitude differ across countries. As shown in the last column of Appendix C, these effects range from 15 per cent in case of India to 50 per cent in case of Iceland, with

 $[\]overline{}^{16} \beta_2 * \Delta distance_{ck,t_0} = 0.152 * (1.471 - 0.482) = 0.15$, which approximately equals 15 per cent, when applying the Taylor expansion in the context of the exponential functional form in the Cox Proportional Hazard Model.

the average effect being slightly above and the median effect being slightly below 30 per cent. 17

4.2 Alternative Channels

Table 4, Table 5, and Table 6 keep the extended set of explanatory variables from columns (3) and (4) of Table 3 and add additional controls capturing alternative financial and non-financial channels that could be correlated with our main mechanism. Similarly to Table 3, the odd-numbered columns report results for the full sample of countries including those countries where domestic stock market was closed or open to foreign investors during the period 1980-1997. The even-numbered columns report results for the subsample of countries where the event of stock market liberalization occurred between 1980 and 1997.

Table 4 examines alternative financial channels that could be correlated with the impact of stock market liberalization on the long-term export survival. Columns (1) and (2) add an interaction between stock market liberalization and dependence on external finance into the set of control variables, with the direct effect of the industry-level dependence on external finance being absorbed by the product strata variable. The additional interaction term controls for the possibility that an improved access to finance benefits in particular those industries that require a high share of external financing for their operations (Rajan and Zingales 1998; Manova 2008). A negative coefficient for this interaction term would confirm that after the event of stock market liberalization the hazard rate in the world market is disproportionately lower for exports from industries dependent on external finance.

Columns (3) and (4) of Table 4 include the interaction term of bank credit to private credit over GDP (capturing the level of domestic banking development in the exporting country) with distance to comparative advantage. This variable controls for the possibility that a strong banking sector disproportionately decreases the survival odds for products that do not correspond to the comparative advantage of the exporting country, in accordance with theories about disciplining role of debt and superior monitoring abilities of banks (Jaud, Kukenova, and Strieborny 2018).

Columns (5) and (6) of Table 4 include the interaction term of stock market capital-

¹⁷ The pattern is similar but number of countries with available data lower when we measure distance to comparative advantage ten years before the liberalization event instead of five years. The results are available upon request.

ization over GDP (capturing the depth of stock markets in the exporting country) with the distance to comparative advantage. This allows for the possibility that the depth of stock markets interacts with the timing of the opening up of the stock market to foreign investors and/or that deep stock markets differentially affect export survival of products depending on their alignment with the comparative advantage of the exporting country.

The results for either the stock market liberalization or its interaction with distance to comparative advantage are not affected by controlling for additional finance channels in Table 4. Both terms maintain their expected sign and statistical significance. As for the alternative channels, the interaction term of stock market liberalization and dependence on external finance is significant with the expected negative sign in columns (1) and (2), confirming the results of Manova (2008) on export volume also in the context of long-term export survival. The interaction term of distance to comparative advantage with the bank development has the expected sign and is statistically significant in columns (3) and (4) while the interaction of distance to comparative advantage with the stock market development in columns (5) and (6) is insignificant, confirming the results of Jaud, Kukenova, and Strieborny (2018) in a different dataset with different set of exporting countries, a different sample period and a different destination market.¹⁸

Table 5 focuses on non-financial alternative mechanisms that could influence the differential impact of stock market liberalization on export dynamics of products that are closer versus products that are further away from the comparative advantage of the exporting country. Regardless of the included additional channels, both stock market liberalization and its interaction with the distance to comparative advantage maintain their significant coefficients with the expected signs throughout Table 5.

Columns (1)-(6) of Table 5 look at the impact of economic development and trade openness in the exporting countries. These two alternative channels are captured by the interaction terms of distance to comparative advantage with GDP per capita and with the sum of exports and imports divided by GDP. Firstly, these additional variables control for the possibility that timing of stock market liberalization is affected by the level of economic development, phases of business cycle or trade openness in the liberalizing countries. Secondly, the levels of economic development and trade openness could affect differently the survival chances of products that are close to the comparative advantage of the exporting country versus products that are far away from it. Columns (1)-(2)

¹⁸ In this paper, we use SITC data that are slightly more aggregated than data used by Jaud, Kukenova, and Strieborny (2018) but allow us to look at a longer time span. Additionally, we look at exports to the whole world market rather than focusing on the US destination market alone.

include the interaction term of GDP per capita with distance to comparative advantage and columns (3)-(4) control for the interaction term of trade openness with distance to comparative advantage. Columns (5)-(6) control for both channels simultaneously. While trade openness seems to decrease the hazard rate overall, there is no evidence for a differentiated effect across products with varying levels of conformity with the comparative advantage. The interaction term between trade openness and distance to comparative advantage is insignificant in columns (3)-(6) where it is included. As for the differentiated impact of countries' level of economic development, the results in Table 5 highlight the importance of controlling for underlying structural differences between countries that opened up to foreign investors already before 1980 and countries that kept their stock market closed until 1997. In columns (1) and (5) that report the results for the whole sample, the interaction term of GDP per capita and distance to comparative advantage has a negative sign. In columns (2) and (4) that report results for the subsample of countries implementing equity market liberalization between 1980 and 1997, the estimated coefficient for the interaction term switches to the more intuitive positive sign, suggesting that products not well aligned with the comparative advantage exit the export portfolio disproportionately faster in countries with a higher level of GDP per capita.

Columns (7) and (8) of Table 5 include the interaction between distance to comparative advantage and the trade liberalization dummy. This interaction term controls for the possibility that the timing of stock market liberalizations might coincide with trade liberalization events, which could then drive our result of a higher post-liberalization hazard rate of the products poorly aligned with the comparative advantage of the liberalizing countries. Similarly to stock market liberalization events or trade openness, trade liberalization events also have a positive direct impact on long-term export performance, decreasing the hazard rate of exports. As for the differentiated effect across products, the interaction between trade liberalization and distance to comparative advantage is statistically significant with the expected positive sign in column (7) of Table 5, suggesting that free-market trade policies lead to disproportionately higher hazard rates for products not well aligned with the comparative advantage. Although this interaction term becomes insignificant in column (8), one needs to stress a certain unfairness of this particular horse race between stock market liberalization and trade liberalization. Trade liberalization is politically less controversial and therefore generally precedes stock market liberalization, often by a significant period of time. Many countries in the subsample of financial liberalizers from column (8) of Table 5 have thus liberalized their trade regime already before 1980 so that the trade liberalization dummy equals one for all export spells from these countries. This might leave insufficient variation to precisely estimate the differential impact of trade liberalization in this specification. In future research, we plan to look deeper into possible interactions and feedback effects between financial and trade liberalization when it comes to eliminating exports of products not corresponding to the comparative advantage of the liberalizing countries.

Table 6 allows various financial and non-financial channels to affect the long-term export survival simultaneously. In columns (1) and (2), we focus on channels that operate through liberalization events capturing sudden changes within countries over time. In particular, we control simultaneously for the differential impact of stock market liberalization on industries requiring a high level of external finance and for the differential effect of trade liberalization across products with varying levels of alignment with the comparative advantage. In columns (3)-(6), we focus on alternative channels that operate through the slowly-moving country characteristics varying mostly across rather than within countries. We thus control for domestic bank or stock market development together with economic development and trade openness, as well as for the potentially differential impact of all those variables across products with different levels of conformity with the comparative advantage. Our main results remain robust throughout the whole Table 6. Allowing foreign investors to acquire equity stakes in domestic firms promotes the long-term presence of exports from liberalizing countries to the world market (i.e., decreases the hazard rate of existing export spells) but it does disproportionately less so for products not well aligned with the comparative advantage of the exporting countries. As for alternative financial mechanisms, the results confirm the importance of stock market liberalizations for industries requiring a high share of external financing (columns (1)-(2)) and the important disciplining role of a well-developed domestic banking sector when it comes to aligning export portfolio of a given country with its comparative advantage (columns (3)-(4)). The general level of domestic stock market development once again fails to exercise the disciplining effect of the stock market liberalization (columns (5)-(6)), highlighting the unique role foreign investors seem to play in this regard. The results for interaction terms of distance to comparative advantage with non-financial variables also remain qualitatively mostly the same as in previous estimations. One exception is the general loss of significance for the interaction between GDP per capita and distance to comparative advantage once we simultaneously control for the alternative financial channels.

5 Robustness Tests

Table 7, Table 8, Table 9, and Online Appendix D provide a series of robustness tests to our main results. For brevity, we report in Tables 7 to 9 only results from the liberalizing subsample that arguably provides a cleaner estimate for our main interaction term due to elimination of cross-country structural differences between mostly rich countries that liberalized their stock market already before 1980 and mostly poor countries that remained closed to foreign investors for the whole time until 1997. The point of departure is thus the specification from column (4) of Table 3. Table 7 adds several control variables that could be correlated with our main interaction term. Table 8 provides a series of econometric robustness tests by employing alternative methods for clustering the standard errors and using different sets of fixed and strata effects. Table 9 addresses several timing issues within the survival framework (time-varying covariates, timing of stock market liberalization in the context of export spells, and an alternative handling of tied-spells termination events). Online Appendix D reports the results of a series of robustness tests using two alternative measures of stock market liberalization, and it includes estimations in both the full sample and the liberalizing subsample.

Column (1) of Table 7 controls for the possibility that the strength of domestic currency affects exports differentially, depending on individual products' alignment with the comparative advantage of the exporting country. We thus add both real exchange rate and its interaction with distance to comparative advantage into the set of control variables. In column (2), we control for the possibility that the entry of foreign investors affect exports of differentiated products differently from the exports of homogeneous products. Here we add interactions of stock market liberalization both with a dummy for differentiated products and with a dummy for products that are reference-priced in trade publications. The omitted dummy is the one for homogeneous products, i.e. the products bought and sold via organized exchanges (see Rauch 1999 for details). The direct effects of both product dummies are captured by the product strata effects. In column (3), we control for the fact that countries receiving a loan from the International Monetary Fund are often obliged to implement various economic reforms that could arguably both improve the overall export performance of the countries and affect the alignment of countries' export portfolio with their comparative advantage (e.g., if IMF-induced reforms affect the corporate governance of the exporting firms). We thus include both an IMF dummy that is equal to one if the exporting country is subject to an IMF program during the duration of a given export spell and an interaction of this dummy with distance to comparative

advantage into our set of control variables. In the first three columns of Table 7, our main interaction term maintains its positive sign and statistical significance. Among the four additional interaction terms, only the interaction between stock market liberalization and dummy for differentiated products included in column (2) is statistically significant. The negative impact of this control variable on the export hazard rate suggests that the entry of foreign investors promotes long-term export survival disproportionately more in the case of differentiated products.

In column (4) and (5) of Table 7, we control for the possibility that the speed with which countries' export portfolios move closer towards their respective comparative advantage varies over time due to various global and country-specific forms of underlying structural change that are unrelated to the process of financial liberalization (technological progress, globalization, processes of information diffusion and/or learning about foreign markets, time-varying corporate procedures, etc.). In column (4), we thus include interaction of distance to comparative advantage with the time trend. As the time is here the year of the initiation of the export spell, the direct effect of the time trend is captured by the time (i.e., spell initiation year) fixed effects. In column (5), we go a step a further and allow for the possibility that trajectories of structural change differ across exporting countries. In particular, we include country-specific time trends and their interactions with distance to comparative advantage into the set of control variables (coefficients not reported for space reasons). In both column (4) and column (5), the estimated coefficient for our main interaction term remains positive and significant. The entry of foreign investors into the domestic stock market thus seems to push export portfolios towards the comparative advantage of the liberalizing countries even after controlling for underlying structural changes captured by the differential impact of global and country-specific time trends.

Table 8 provides a series of econometric robustness tests related to alternative methods for clustering the standard errors, stratification, and the use of fixed effects. For better comparison, the first column reports the estimation from the column (4) of Table 3. In second column, we cluster standard errors at the level of exporting country rather than at the country*time level, addressing the econometric issues raised by Bertrand, Duflo, and Mullainathan (2004). In the third column, we cluster the standard errors simultaneously alongside both exporting-country and time dimension, following the two-way procedure suggested by Cameron, Gelbach, and Miller (2006). As the only difference among these three estimations relates to the way how the standard errors are computed, the point estimates are the same in columns (1)-(3) of Table 8. In columns (4)-(7) of Table 8, we use alternative versions of fixed and strata effects (for the difference between the two, see Appendix B2). In the fourth column, we replace the product stratification by the product*time stratification. This approach controls for all observable and unobservable effects that could vary both across products and over time, like the level of competition for a given product in the world market. Our proxy for such competition (number of suppliers) is thus absorbed by these more stringent strata effects in column (4) of Table 8. In the fifth column, we apply the product*country stratification that controls for various trade policies at product-country level (e.g., export subsidies) that could affect longterm survival of individual products from individual countries in the world market. In the sixth column, we replace separate time and exporting-country fixed effects by the interacted (exporting-country)*time fixed effects. This set of fixed effects controls for all influences that vary both across exporting countries and over time and consequently absorbs the direct effect of several control variables (total exports, GDP per capita, factor endowments).¹⁹ In the seventh column, we take the contrary approach to columns (4)-(6)where we applied stricter sets of strata and fixed effects, and we instead refrain from the use of any strata or fixed effects. In all columns of Table 8, our main interaction term remains positive and significant.

In Table 9, we address several issues of timing that arise in the context of examining the effects of stock market liberalization events in the econometric framework of longterm export survival. In the first column, we allow for the possibility that the effect of our main variables of interest is not constant but gets stronger or weaker over the export spells' lifetimes.²⁰ The direct effect of stock market liberalization seems indeed to be time-varying. Allowing foreign investors to acquire equity stakes in domestic firms decreases the overall hazard rate of exports from the liberalizing country to the world market (negative and significant coefficient for stock market liberalization in the first subcolumn of column (1) in Table 9) and this effect gets stronger over time (negative and significant coefficient for stock market liberalization of column (1) in Table 9). Our main interaction term maintains its overall positive and significant

¹⁹ By contrast, the (exporting-country)*time fixed effects do *not* absorb the direct effect of stock market liberalization. The liberalization dummy is not measured in the year when a given export spell started but instead equals one if the liberalization event occurred before or during the spell. This variable is thus spell-specific instead of varying solely across countries and time.

²⁰ The use of time-varying covariates in the Cox survival framework prevents convergence in the presence of large sets of fixed effects. In the first column of Table 9, we therefore drop time and exportingcountry fixed effects and keep only stratification at the product level.

coefficient (first sub-column) and this effect seems to be constant, getting neither stronger nor weaker over the duration of export spells (insignificant coefficient in the second subcolumn).²¹ In the second column of Table 9, we follow the export spells only until 1997 rather than 2006, unifying the time spans for stock market liberalization dates and export spells. In the third column of Table 9, we drop those observations where stock market liberalization occurs during the lifetime of a given export spell, i.e. we drop those export spells that were only "partially treated" by the ability of foreign investors to acquire equity stakes in domestic firms. In the fourth column of Table 9, we use the Efron method (Efron 1988) to handle tied-spells termination events.²²

In Online Appendix D, we perform a series of robustness tests using the dummy for the first sign of liberalization and a continuous variable capturing the intensity of openness of domestic stock market towards foreign investors from Bekaert, Harvey, and Lundb-land (2005) as two alternative measures of stock market liberalization. The liberalization dummy used throughout the main text is based on the official year of stock market liberalization that corresponds to the date of a formal legal change allowing foreign investors to acquire equity stakes in domestic firms. The first sign of liberalization corresponds to the earliest of the following three dates in the liberalizing country: the year of official liberalization, the year of the first American Depository Receipt (ADR, a security allowing the shares of non-US companies to be traded in the US financial markets), launch of the first country fund (a fund with portfolio containing only stocks of a given country). A continuous variable capturing the intensity of stock market liberalization is representing the share of domestic stocks available to foreign investors.

The results for our main interaction term are generally stronger in Online Appendix D than in the main text (e.g., comparing the significance level for our main interaction term in columns (5) and (6) of Table VI and Table VII in Online Appendix D with columns (5) and (6) of Table 6 in the main text). This might be caused by two possible sources of a bias against finding significant results in our difference-in-difference framework when using a stock market liberalization dummy based on the official stock market liberalization dates. Firstly, foreign investors might acquire equity stakes in domestic firms via ADR or equity funds even before the official dates of equity market liberalization.

²¹ Appendix B3 provides technical background for this interpretation.

²² Tied-spells termination events occur when two or several export spells end at the same time, violating one of the assumptions of the continuous-time Cox Proportional Hazard Model. In other estimations, we are using instead the standard Breslow method (Breslow 1974) that requires substantially less computing time.

The official liberalization dummy thus underestimates the influence of foreign investors before the liberalization event. Secondly, even after the government in principle allows foreign investors to acquire equity stakes in domestic firms, it might still keep in place certain restrictions on maximum foreign ownership or foreign voting rights. The official liberalization dummy thus overestimates the influence of foreign investors after the liberalization event. Consequently, the real difference in the ability of foreign investors to affect the country's export portfolio in the time before versus the time after the equity market liberalization event is smaller than a binary dummy based on the official liberalization date would suggest. The use of such dummy thus biases the coefficient for the interaction term of liberalization dummy with distance to comparative advantage towards zero. Using instead the dummy for the first sign of liberalization or a continuous variable capturing the liberalization intensity address this issue to a certain extent although a possible measurement error might still generate a bias towards zero.²³ In that sense, the results for our main interaction term reported in this paper probably represent only a lower bound for the true impact of foreign investors on pushing the export portfolios of countries towards their comparative advantage.

6 Monitoring by Foreign Investors

The results presented in Section 4 and Section 5 show that the entry of foreign investors into domestic stock market triggers an efficiency-enhancing structural change in the real economy by re-balancing countries' export portfolios towards their comparative advantage. A possible explanation for this result might lie in the monitoring and disciplining effects of foreign investors in domestic firms. An influential strand of literature argues that foreign investors are more effective as external monitors because they lack business relations with local firms that often induce domestic investors to feel loyal to the local management (e.g., Gillan and Starks 2003; Ferreira and Matos 2008; Aggarwal et al. 2011; Bena et al. 2017; Luong et al. 2017). If that is true, then stricter monitoring by foreign investors should push the exporting managers in domestic firms to focus on products that correspond to the comparative advantage of the domestic economy and abandon those

²³ Different definitions of availability of domestic equity to foreign investors make it difficult to measure stock market liberalization intensity exactly and unequivocally. For instance, Bekaert, Harvey, and Lundbland (2005) base their measure of liberalization intensity on ratios of IFC Investable Index and IFC Global Index and report an average liberalization intensity for Finland during the 1980-1997 period that is equal to one, i.e. 100 per cent. By contrast, Liljeblom and Löflund (2005) report that foreign ownership of companies in Finland was restricted to 40 per cent of equity and 20 per cent of voting rights prior to 1993.

products that make suboptimal use of the available resources in the domestic economy. This section provides some preliminary evidence in accordance with this interpretation.²⁴

If our findings are indeed driven by the superior monitoring skills of foreign investors, then the results should become stronger in a legal environment providing sufficient protection to the rights of investors in general and of foreign investors in particular. Arguably, foreign investors can fully employ their monitoring abilities only if they can rely on a robust legal framework protecting their rights vis-vis corporate insiders and local government. Table 10 examines this hypothesis, based on a composite index Legal System and Property Rights from the Economic Freedom Database. The index captures various legal dimensions relevant for investor protection like integrity of legal system, legal enforcement of contracts, protection of property rights, impartiality of courts, judicial independence, etc., with higher values of the index corresponing to higher levels of investor protection. The index varies both across countries and over time. Due to possible endogeneity concerns, we use the values measured in 1980, which is the first year of our sample.

Column (1) of Table 10 runs our preferred specification for all countries from column (3) of Table 3 on the subsample of countries for which we have data on the Legal System and Property Rights variable. The number of observations thus drops by approximately 10 per cent from 85,675 to 77,165. In columns (2) and (3), we split this sample into countries below and above median of our variable capturing the legal environment in exporting countries. Our main interaction term is only marginally significant in the subsample of countries with a below-average legal system reported in column (2). By contrast, the main interaction term is highly significant in the subsample of countries with an above-average legal system reported in column (3). The size of the coefficient for our main interaction term is also substantially higher in the third column compared to the second column of Table 10. In columns (4)-(6), we repeat the exercise on the subsample of liberalizing countries that allowed foreign investors to acquire equity stakes in domestic firms for the first time at some point during the 1980-1997 period. The point of departure is thus column (4) of Table 3, with column (4) of Table 10 re-running the estimation on those liberalizing countries for which we have data on the Legal System and Property Rights variable. The number of observations now drops by around 7 per cent, from 45,699

²⁴ Efficiency-maximizing managers would abandon of their own volition those exports that use the available resources suboptimally. However, exporting activities might generate substantial managerial perks that would induce self-interested managers to keep alive also those exports that they know to be inefficient. Jaud, Kukenova and Strieborny (2018) present a formal model alongside these lines.

to 42,376, suggesting that we have the legal data for the vast majority of the liberalizing countries. Columns (5) and (6) report the results for liberalizing countries below and above median of the Legal System and Property Rights variable.²⁵ The difference is even more pronounced than in the whole sample. In column (5), our main interaction term is insignificant. In column (6), the main interaction is highly significant with the size of its coefficient increasing more than tenfold compared to column (5). The results reported in previous sections thus seem to be driven solely by those liberalizing countries that had a sufficient level of investor protection in place already before allowing foreign investors to enter their domestic stock markets.²⁶

In Online Appendix E, we provide a series of robustness tests by re-running Table 10 with sample splits based on alternative legal variables used by La Porta et al. (1998). These include Efficiency of Judicial System (an assessment of the "efficiency and integrity of the legal environment as it affects business, particularly foreign firms" produced by the country-risk rating agency Business International Corp.), Rule of Law (an assessment of the law and order tradition in the country produced by the country-risk rating agency International Country Risk (ICR)), Risk of Expatriation (an ICR's assessment of the risk of "outright confiscation" or "forced nationalization"), Repudiation of Contracts by Government (an ICR's assessment of the "risk of modification in a contract taking the form of repudiation, postponement or scaling down" due to "budget cut-back, indigenization pressure, a change in government, or change in government economic and social priorities"), and Investor Protection (not used by La Porta et al. 1998, it is an average of the three ICR measures computed by us).

Some of these alternative variables might arguable represent an even better proxy for protection of investors in general and of foreign investors in particular. We have decided to report the results based on the Legal System and Property Rights variable in the main text because it has the best country coverage yielding the highest number of observations,

²⁵ Note that the minimum thresholds identifying countries as "above average" in terms of Legal System and Property Rights variable are different in the full sample and the subsample of countries that experienced stock market liberalization during the 1980-1997 period. We have also re-run the estimations in columns (5) and (6) of Table 10 based on the threshold used in columns (2) and (3) of Table 10. The results are available upon request.

²⁶ Note that we obtain stronger evidence for a differential impact of stock market liberalization in countries with an above-average legal system (third and sixth column of Table 10) despite a significantly lower number of observations in this subsample compared to the subsample of countries with a below-average legal system (second and fifth column of Table 10). The difference in number of observations is due to fact that our sample split is based on the country-level variable Legal System and Property Rights. For instance, countries with an above-average legal system might have fewer but longer export spells compared to countries with a below-average legal system.

and because it is measured at the beginning of our sample (1980) rather than representing the averages over the years 1980-1983 or 1982-1995 (see la Porta et al. 1998 for more details). If anything, the results based on the alternative legal measures are even stronger than the results reported in Table 10. For example, when it comes to the specification reported in column (2) based on the subsample of countries with below-average values of legal variables, our main interaction term is not significant at even 10 per cent level in the tables reported in Online Appendix E.

7 Conclusions

This paper provides evidence that allowing foreign investors to acquire equity stakes in domestic firms significantly improves the chances of domestic products to establish a successful long-term presence in the world market. However, this beneficial effect is not uniform across products but depends on the extent to which a given product corresponds to the comparative advantage of the country that opens up its stock markets to foreign investors. The further away a product is from the comparative advantage of the liberalizing country, the sooner it exits this country's export portfolio after the stock market liberalization event occurs.

Our results suggests that the entry of foreign investors triggers an efficiency-enhancing structural change, by pushing the recipient countries' export portfolios towards the products that make the optimal use of available resources. While we find strong evidence for this disciplining effect on export composition following the opening of stock markets to foreign investors, we find no such effect when looking at the standard proxy for domestic stock market development. The uniqueness of foreign investors in this regard seems to be in accordance with the existing literature that makes the case for superior monitoring abilities of foreign investors (Gillan and Starks 2003; Ferreira and Matos 2008; Aggarwal et al. 2011; Bena et al. 2017; Luong et al. 2017). Section 6 provides some preliminary evidence consistent with this interpretation.

Future research could build upon evidence presented in Section 6 and further explore the specific transmission channels from foreign investors to structural change in the real economy. One research venue could be connected to the renewed academic interest in "voice" (direct actions through board memberships, voting or behind-the scenes interventions) and "exit" (selling their equity stake or a part of it) as alternative mechanisms through which investors shape the actions of firms whose shares they hold. While recent literature examined these two mechanisms mostly in the common-law context of US shareholders and firms (see Edmans and Holderness 2017 for a recent survey), exploring them in the cross-border context of foreign investors and domestic firms could provide new insights. Foreign investors might namely choose different paths when it comes to voice versus exit, depending among other things on the legal and regulatory framework in both their own countries and the countries where they invest. Another venue for future research could lie in the interactions between foreign and domestic investors. For instance, the entry of foreign investors to an hitherto closed domestic stock market would arguably increase the liquidity of such market, improving the efficiency of exit strategy also for the existing domestic shareholders, alongside the lines of Edmans and Manso (2011). By contrast, the impact of increased liquidity on voice strategy of existing shareholders could be either positive (e.g., Maug 1998) or negative (e.g., Bhide 1993; McCahery, Sautner, and Starks 2016).

Naturally, a rigorous analysis of specific transmission mechanisms from foreign investors to structural change cannot rely on product-level data alone. This kind of analysis would also require the use of firm-level data, and it would therefore need to address the issue of spillovers and economic externalities associated with foreign investors discussed in the Introduction. The limited availability of internationally comparable firm-level data would also imply a reduced country coverage of such empirical investigation.

References

- Aggarwal, Reena, Isil Erel, Miguel Ferreira, and Pedro Matos, 2011, "Does Governance Travel around the World? Evidence from Institutional Investors," Journal of Financial Economics 100, pp. 154-181.
- Albuquerque, Rui, Luis Brandao-Marques, Miguel Ferreira, and Pedro Matos, 2019, "International Corporate Governance Spillovers: Evidence from Cross-Border Mergers and Acquisitions," Review of Financial Studies 32(2), pp. 738-770.
- Amiti, Mary, and David E. Weinstein, 2011, "Exports and Financial Shocks," Quarterly Journal of Economics 126(4), pp. 1841–1877.
- Aslan, Hadiye, and Praveen Kumar, 2016, "The Product Market Effects of Hedge Fund Activism," Journal of Financial Economics 119, pp. 226–248.

- Balassa, Bela, 1965, "Trade Liberalisation and "Revealed" Comparative Advantage," Manchester School 33(2), pp. 99-123.
- Bhagwati, Jagdish N., 1972, "The Heckscher-Ohlin Theorem in the Multi-Commodity Case," Journal of Political Economy 80(5), pp. 1052-1055.
- Beck, Thorsten, 2002, "Financial Development and International Trade: Is There a Link?" Journal of International Economics 57, pp. 107–131.
- Beck, Thorsten, 2003, "Financial Dependence and International Trade," Review of International Economics, 11(2), pp. 296-316.
- Beck, Thorsten, Asli Demirguc-Kunt, and Ross Levine, 2000, "A New Database on Financial Development and Structure," World Bank Economic Review 14, pp. 597-605.
- Beck, Thorsten and Ross Levine, 2002, "Industry Growth and Capital Allocation: Does Having a Market– or Bank–Based System Matter?," Journal of Financial Economics 64, pp. 147–180.
- Becker, Bo, Jinhzu Chen, and David Greenberg, 2013, "Financial Development, Fixed Costs, and International Trade," Review of Corporate Finance Studies 2(1), pp. 1–28.
- Bena, Jan, Miguel Ferreira, Pedro Matos and Pedro Pires, 2017, "Are Foreign Investors Locusts? The Long-Term Effects of Foreign Institutional Ownership," Journal of Financial Economics 126, pp. 122-146.
- Bekaert, Geert, and Campbell R. Harvey, 2000, "Foreign Speculators and Emerging Equity Markets," Journal of Finance 55(2), pp. 565-613.
- Bekaert, Geert, Campbell R. Harvey, and Christian Lundblad, 2005, "Does Financial Liberalization Spur Growth?" Journal of Financial Economics 77, pp. 3-35.
- Bekaert, Geert, Campbell R. Harvey, Christian Lundblad, and Stephan Siegel, 2007, "Global Growth Opportunities and Market Integration" Journal of Finance 62(3), pp. 1081-1137.
- Bertrand, Marianne, Esther Duflo, and Sendhil Mullainathan, 2004, "How Much Should We Trust Differences-in-Differences Estimates?" Quarterly Journal of Economics 119(1), pp. 249-275.

- Besedes, Tibor, and Thomas J. Prusa, 2006a, "Ins, Outs, and the Duration of Trade," Canadian Journal of Economics 39(1), pp. 266-295.
- Besedes, Tibor, and Thomas J. Prusa, 2006b, "Product Differentiation and Duration of US Import Trade," Journal of International Economics 70, pp. 339-358.
- Besedes, Tibor, and Thomas J. Prusa, 2011, "The Role of Extensive and Intensive Margins and Export Growth," Journal of Development Economics 96, pp. 371-379.
- Bhide, Amar, 1993, "The Hidden Costs of Stock Market Liquidity," Journal of Financial Economics 34(1), pp. 31–51.
- Bradley, Michael, Alon Brav, Itay Goldstein, and Wei Jiang, 2010, "Activist Arbitrage: A Study of Open-Ending Attempts of Closed-End Funds," Journal of Financial Economics 95, pp. 1–19.
- Braun, Matías, 2005, "Financial Contractibility and Asset Hardness," Working Paper.
- Brenton, Paul, Christian Saborowski, and Erik von Uexkull, 2010, "What Explains the Low Survival Rate of Developing Country Export Flows?" World Bank Economic Review 24(3), pp. 474–499.
- Breslow, Norman, 1974, "Covariance Analysis of Censored Survival Data." Biometrics, pp. 89-99.
- Broner, Fernando, and Jaume Ventura, 2016, "Rethinking the Effects of Financial Globalization," Quarterly Journal of Economics 131(3), pp. 1497-1542.
- Cadot, Olivier, Celine Carrere, and Vanessa Strauss–Kahn, 2011, "Export Diversification: What's Behind the Hump?" Review of Economics and Statistics 93(2), pp. 590–605.
- Cadot, Olivier, Leonardo Iacovone, Martha D. Pierola, and Ferdinand Rauch, 2013, "Success and Failure of African Exporters," Journal of Development Economics 101, pp. 284–296.
- Cadot, Olivier, Bolormaa Tumurchudur, and Miho Shirotori, 2009, "Revealed Factor Intensity Indices at the Product Level," UNCTAD.
- Cameron, Colin A., Jonah B. Gelbach, and Douglas L. Miller, 2006, "Robust Inference with Multi-Way Clustering," NBER Technical Working Paper 327.

- Deardorff, Alan V., 1979, "Weak Links in the Chain of Comparative Advantage," Journal of International Economics 9, pp. 197-209.
- Defever, Fabrice, and Jens Suedekum, 2014, "Financial Liberalization and the Relationship-Specificity of Exports," Economics Letters 122, pp. 375-379.
- Doidge, Craig, Andrew Karolyi and René Stulz, 2007, "Why Do Countries Matter So Much for Corporate Governance?" Journal of Financial Economics 86, pp. 1-39.
- Edmans, Alex, Itay Goldstein, and Wei Jiang, 2012, "The Real Effects of Financial Markets: The Impact of Prices on Takeovers," Journal of Finance 67(3), pp. 933–971.
- Edmans, Alex, and Clifford G. Holderness, 2017, "Blockholders: A Survey of Theory and Evidence," in Benjamin Hermalin and Michael Weisbach (eds): Handbook of the Economics of Corporate Governance, North Holland, Amsterdam, Netherlands, pp. 541-636.
- Edmans, Alex, and Gustavo Manso, 2011, "Governance Through Trading and Intervention: A Theory of Multiple Blockholders," Review of Financial Studies 24(7), pp. 2395–2428.
- Efron, Bradley, 1988, "Logistic Regression, Survival Analysis, and the Kaplan-Meier Curve," Journal of the American Statistical Association 83(402), pp. 414-425.
- Fisman, Raymond and Inessa Love, 2004, "Financial Development and Intersectoral Allocation: A New Approach," Journal of Finance 59(6), pp. 2785–2807.
- Fisman, Raymond and Inessa Love, 2007, "Financial Dependence and Growth Revisited," Journal of the European Economic Association 5(2-3), pp. 470–479.
- Ferreira, Miguel, and Pedro Matos, 2008, "The Colors of Investors' Money: The Role of Institutional Investors around the World," Journal of Financial Economics 88, pp. 499-533.
- Gillan, Stuart, and Laura Starks, 2003, "Corporate Governance, Corporate Ownership, and the Role of Institutional Investors: A Global Perspective," Journal of Applied Finance 13, pp. 4-22.
- Gantchev, Nickolay, Oleg Gredil, and Chotibhak Jotikasthira, 2018, "Governance under the Sun: Spillover Effects of Hedge Fund Activism," ECGI Working Paper 562/2018.

- Gupta, Nandini, and Kathy Yuan, 2009, "On the Growth Effect of Stock Market Liberalizations," Review of Financial Studies 22(11), pp. 4715-4752.
- Gwartney, James D., Robert Lawson, and Seth Norton, 2008, "Economic Freedom of the World Annual Report," The Fraser Institute.
- Hausmann Ricardo, Jason Hwang, and Dani Rodrik, 2007, "What You Export Matters," Journal of Economic Growth 12(1), pp. 1-25.
- Henry, Peter Blair, 2000a, "Do Stock Market Liberalizations Cause Investment Booms?" Journal of Financial Economics 58, pp. 301-334.
- Henry, Peter Blair, 2000b, "Stock Market Liberalization, Economic Reform, and Emerging Market Equity Prices," Journal of Finance 55(2), pp. 529-564.
- Henry, Peter Blair, 2006, "Capital Account Liberalization: Theory, Evidence, and Speculation," Journal of Economic Literature 45(4), pp. 887-935.
- Jaud, Melise, Madina Kukenova, and Martin Strieborny, 2015, "Financial Development and Sustainable Exports: Evidence from Firm-product Data," World Economy 38(7), pp. 1090-1114.
- Jaud, Melise, Madina Kukenova, and Martin Strieborny, 2018, "Finance, Comparative Advantage, and Resource Allocation," Review of Finance 22(3), pp. 1011-1061.
- Javorcik, Beata Smarzynska, 2004, "Does Foreign Direct Investment Increase the Productivity of Domestic Firms? In Search of Spillovers Through Backward Linkages," American Economic Review 94(3), pp. 605-627.
- Jones, Ronald W., 1956-1957, "Factor Proportions and the Heckscher-Ohlin Theorem," Review of Economic Studies 24(1), pp. 1-10.
- Kaminsky, Graciela Laura, and Sergio L. Schmukler, 2008, "Short-Run Pain, Long-Run Gain: Financial Liberalization and Stock Market Cycles," Review of Finance 12(2), pp. 253-292.
- La Porta, Rafael, Florencio Lopez-de-Silanes, Andrei Shleifer, and Robert W. Vishny, 1998, "Law and Finance," Journal of Political Economy 106(6), pp. 1113-1155.

- Love, Inessa, 2003, "Financial Development and Financing Constraints: International Evidence from the Structural Investment Model," Review of Financial Studies 16(3), pp. 765-791.
- Larrain, Mauricio, and Sebastian Stumpner, 2017, "Capital Account Liberalization and Aggregate Productivity: The Role of Firm Capital Allocation" Journal of Finance 72(4), pp. 1825-1857.
- Levchenko, Andrei, Romain Ranciere, and Mathias Thoenig, 2009, "Growth and Risk at the Industry Level: The Real Effects of Financial Liberalization," Journal of Development Economics 89(2), pp. 210-222.
- Liljeblom, Eva, and Anders Löflund, 2005, "Determinants of International Portfolio Investment Flows to a Small Market: Empirical Evidence," Journal of Multinational Financial Management 15, pp. 211-233.
- Luong, Hoang, Fariborz Moshirian, Lily Nguyen, Xuan Tian, and Bohui Zhang, 2017, "How Do Foreign Institutional Investors Enhance Firm Innovation?" Journal of Financial and Quantitative Analysis 52(4), pp. 1449-1490.
- Manova, Kalina, 2008, "Credit Constraints, Equity Market Liberalizations and International Trade," Journal of International Economics 76, pp. 33-47.
- Manova, Kalina, 2013, "Credit Constraints, Heterogeneous Firms, and International Trade," Review of Economic Studies 80, pp. 711–744.
- Maug, Ernst, 1998, "Large Shareholders as Monitors: Is There a Trade–Off between Liquidity and Control?," Journal Finance 53(1), pp. 65-98.
- McCahery, Joseph, Zacharias Sautner, and Laura T. Starks, 2016, "Behind the Scenes: The Corporate Governance Preferences of Institutional Investors," Journal of Finance 71(6), pp. 2905–2932.
- Nitsch, Volker, 2009, "Die Another Day: Duration in German Import Trade," Review of World Economics 145(1), pp. 133–154.
- Paravisini, Daniel, Veronica Rappoport, Philipp Schnabl, and Daniel Wolfenzon, 2015,
 "Dissecting the Effect of Credit Supply on Trade: Evidence from Matched Credit-Export Data," Review of Economic Studies 82, pp. 333–359.

- Rajan, Raghuram G., and Luigi Zingales, 1998, "Financial Dependence and Growth", American Economic Review 88(3), pp. 559-86.
- Rauch, James E., 1999, "Networks versus Markets in International Trade," Journal of International Economics 48, pp. 7-35.
- Reinhart, Carmen, and Kenneth Rogoff, 2009, This Time Is Different: Eight Centuries of Financial Folly. Princeton University Press.
- Sachs, Jeffrey D., and Andrew Warner, 1995. "Economic Reform and the Process of Global Integration." Brookings Papers on Economic Activity 1, pp. 1–118.
- Strieborny, Martin, 2017, "Suppliers, Investors, and Equity Market Liberalizations," mimeo, Lund University.
- Strieborny, Martin, and Madina Kukenova, 2016, "Investment in Relationship-Specific Assets: Does Finance Matter?" Review of Finance 20(4), pp. 1487-1515.
- Stulz, René, 2005, "The Limits of Financial Globalization," Journal of Finance 60(4), pp. 1595-1636.
- Varela, Liliana, 2018, "Reallocation, Competition, and Productivity: Evidence from a Financial Liberalization Episode," Review of Economic Studies 85, pp. 1279-1313.
- Wacziarg, Romain, and Karen Horn Welch, 2008, "Trade Liberalization and Growth: New Evidence," World Bank Economic Review, 22(2), pp.187-231.
- Wurgler, Jeffrey, 2000, "Financial Markets and the Allocation of Capital," Journal of Financial Economics 58, pp. 187–214.

Appendix A: Stock Market Liberalizations - Country List (some of the countries below do not enter our regressions due to the lack of data for some of the control variables)

Domestic stock market opened up to foreign investors during 1980-1997: Argentina, Bangladesh, Brazil, Chile, Colombia, Cote d'Ivoire, Ecuador, Egypt, Ghana, Greece, Iceland, India, Indonesia, Israel, Jamaica, Japan, Jordan, Kenya, Malaysia, Malta, Mauritius, Mexico, Morocco, New Zealand, Nigeria, Oman, Pakistan, Peru, Philippines, Portugal, Saudi Arabia, South Africa, South Korea, Spain, Sri Lanka, Thailand, Trinidad and Tobago, Tunisia, Turkey, Venezuela, and Zimbabwe.

Domestic stock market opened up to foreign investors prior to 1980: Australia, Austria, Barbados, Canada, Denmark, Finland, France, Germany, Ireland, Italy, Netherlands, Singapore, Sweden, Switzerland, United Kingdom, and United States

Domestic stock market remained closed to foreign investors until 1997: Algeria, Benin, Burkina Faso, Cameroon, Central African Republic, Chad, Congo, Costa Rica, Dominican Republic, El Salvador, Fiji, Gabon, Gambia, Guatemala, Guyana, Haiti, Honduras, Iran, Kuwait, Madagascar, Malawi, Mali, Nepal, Nicaragua, Niger, Norway, Paraguay, Rwanda, Senegal, Sierra Leone, Syria, Togo, Uruguay, and Zambia.

Appendix B1: The Cox Proportional Hazard Model

The duration of a country export for a given product is defined as the time (measured in years) when a trade relationship has been in existence without interruption. The distribution of durations can be characterized in terms of the hazard function which is defined as the instantaneous probability that a trade relationship ends at time t given that it has survived until time t-1:

$$h(t|X) = Pr(T = t|T \ge t, X), t = 1, 2, \dots$$
(3)

The Cox Proportional Hazard Model assumes that the hazard rate is the product of a unspecified baseline hazard function, which depends only on time, and the exponential function of the covariates:

$$h(t, x, \beta) = h_0(t) \exp(X'\beta) \tag{4}$$

This specification assumes that the covariates affect the hazard function independently on time the trade relationship exists, shifting by the same magnitude all points of a baseline function. The coefficients can be interpreted as semi-elasticities, as they measure the effect of a change in the right-hand side variables on the log of the hazard rate. Due to this structure, the Cox Proportional Hazard Model is very versatile and can fit various models. The baseline hazard can vary across different groups, called strata, but the vector of parameters is restricted to be the same:

$$h_j(t, x, \beta) = h_{j0}(t) \exp(X'\beta)$$
(5)

Since the baseline hazard function remains unspecified, only the order of duration provides information about the unknown parameters. The model is estimated by maximizing a partial likelihood function with respect to the vector of parameters β without specifying the form of the baseline hazard function $h_0(t)$. The estimated parameters reflect the relationship between the explanatory variables and the hazard function (i.e., the risk for a trade relationship to end).

There are several issues related to the duration analysis which need to be addressed. First, observations may be right-censored. This is the case when trade relationships are still in progress in the final year of the sample period. The Cox Proportional Hazard Model can handle right-censored observations. Second, observations may be left-censored, which means that we cannot determine the date when they were initiated. In this situation, the actual length of the spells cannot be determined. To mitigate this problem, we estimate the model after dropping the left-censored observations, that is, the observations for which trade flows were recorded before 1980. Third, some trade relationships may have several periods of continuous exporting (spells). An exporter can enter the market, export for a while, exit and re-enter again. Such consecutive exits may be interrelated. The first exit may increase probability of the following exits. To account for this issue, we introduce a variable "number of spells" indicating the number of trade relationships for every countryproduct pair.

Appendix B2: Fixed Effects versus Stratification in the Cox Proportional Hazard Model

In a non-stratified Cox Proportional Hazard (PH) Model, all export spells would share a common underlying hazard rate $h_0(t)$. In a stratified Cox PH Model, the baseline hazard rate is allowed to vary across different groups (strata). Let us define strata at the product level, allowing export spells of different products k to have different underlying hazard rate $h_{0,k}(t)$. The overall hazard rate of product k would then write:

$$h(t|X) = h_{0,k}(t) \exp(X.\beta).$$

Let us now use product fixed effects instead. This means keeping underlying hazard rate uniform across products but including among regressors a set of dummy variables that are equal to one if the observation belongs to product k and zero otherwise. The overall hazard rate of product k would then write:

$$h(t|X) = h_0(t) \exp(X \cdot \beta + \alpha_k D_k),$$

with α_k being the estimated coefficient for the dummy variable D_k and dummy variable D_k itself being equal to one for any export spell related to product k.

This expression can be rewritten as:

$$h(t|X) = h_0(t) \exp(X.\beta) \exp(\alpha_k D_k) = h_{0,k}(t) \exp(X.\beta),$$

with $\overline{h_{0,k}(t)} \equiv h_0(t) \exp(\alpha_k D_k)$.

Using fixed effects can be thus interpreted as a particular case of stratification, which assumes that baseline hazard rates differ across products merely by the factor of proportionality.

Appendix B3: The Cox Proportional Hazard Model with Time-Varying Covariates

In the Cox Proportional Hazard Model, covariates have a proportional effect on hazard rate and do not change over the spell life:

$$h(t, x, \beta) = h_0(t) \exp(X'\beta).$$

We can relax this assumption by using the extended (time-dependent) Cox model which incorporates time-varying effects of specific covariates by interacting these covariates with some known functions of time:

$$h(t, x, \beta) = h_0(t) \exp(X' * g(\beta, t)),$$

where $g(\beta, t)$ is a specific function of time. If $g(\beta, t)$ is a simple function, it can be written as $g(\beta, t) = \beta * g(t)$.

Stata allows us to estimate the extended (time-dependent) Cox model by adding to the standard command *stcox* the option *tvc* to define time-varying covariates and the option *texp* to define the functional form for g(t). More specifically, we would like to allow our three main variables of interest (distance to comparative advantage, stock market liberalization dummy, and their interaction term) to have time-varying effects on hazard rate. For $g_i(t)$ we use a default option, setting $g_i(t)$ to be a linear function of time: $g_i(t) = \beta_i + \gamma_i * t$

$$h_{ck}(t) = h_{0,k}(t) \exp[g_1(t)StMLib_{c,t_0} + g_2(t)StMLib_{c,t_0} * distance_{ck,t_0} + g_3(t) * distance_{ck,t_0} + \mathbf{Y}_{ck,t_0}\phi + \varepsilon_{ck,t_0}]$$

In the first sub-column of column (1) in Table 9, we present coefficients β_i (our standard Cox model coefficients). In the second sub-column of column (1) in Table 9, we present coefficients γ_i (the coefficients for time-varying components). For a given independent variable, significant coefficients with the same (opposite) signs in both sub-columns imply that the effect of the variable on the hazard rate gets stronger (weaker) over time. An insignificant coefficient in the second sub-column suggest that the effect of a given variable is constant over time.

Country	Year when DCA measured	Date of StM Lib	DCA value (top 25%)	DCA value (bottom 25%)	Difference DCA	Coef(Lib x DCA) x Difference DCA =0.152 x Difference DCA
Arcentine	108/	1080	4.049	1 919	2 836	0.431
Rangladesh	1984	1905	2.043	0.858	1.000	0.185
Brazil	1986	1991	1 689	0.493	1 196	0.182
Chile	1987	1991	2.995	0.305	2 690	0.409
Colombia	1986	1992	2.995	0.139	2.050 2.856	0.434
Ecuador	1989	1994	2.330 2.249	0.113	2.000 2.134	0.324
Egypt	1987	1992	2.243 2.007	0.114 0.214	1.793	0.324 0.273
Ghana	1988	1992	1 874	0.669	1.700 1 205	0.183
Greece	1982	1995	4 014	0.867	3.147	$0.100 \\ 0.478$
Indonesia	1984	1989	2 466	0.007	2 253	0.343
India	1987	1002	2.400	0.482	0.990	0.150
Icoland	1086	1001	1.471	1 360	3 285	0.100
Icerand	1980	1003	4.040	1.300 1.169	3.280	0.435
Ismaias	1988	1995	4.555 2.708	0.686	0.130 0.110	0.400
Jamaica	1980	1991	2.798	0.080	2.112 1 593	0.321 0.221
Joruan	1990	1995	1.709	0.200	1.525	0.231
Kenya Koroa(ropublic of)	1990	1995	1.000 2.722	0.298	1.000 2.820	0.241 0.432
Sri Lanka	1907	1992	3.722 2.001	0.005	2.039	0.432
Marrian	1960	1991	2.091	0.407	1.004	0.244 0.271
Melte	1964	1969	2.102 2.575	0.370	1.765	0.271
	1987	1992	2.070	0.734	1.041	0.280
	1969	1994	2.910	0.198	2.712	0.412 0.167
Malaysia	1983	1988	1.474	0.370	1.098	0.167
New Zealand	1982	1987	3.791	1.312	2.480	0.377
Pakistan	1986	1991	1.875	0.120	1.755	0.267
Peru	1987	1992	2.459	0.122	2.337	0.355
Philippines	1986	1991	3.078	0.594	2.484	0.378
Portugal	1981	1986	2.430	0.458	1.972	0.300
Thailand	1982	1987	1.858	0.177	1.681	0.255
Trinidad & Tobago	1992	1997	2.360	0.679	1.680	0.255
Tunisia	1990	1995	2.340	0.642	1.698	0.258
Turkey	1984	1989	2.847	0.404	2.444	0.371
Venezuela	1985	1990	2.689	0.423	2.266	0.344
South Africa	1991	1996	2.120	0.382	1.738	0.264
Zimbabwe	1988	1993	1.934	0.073	1.860	0.283
Mean Magnitude Median Magnitude						$egin{array}{l} 0.314\ (\%)\ 0.291\ (\%) \end{array}$

Appendix C: Economic Magnitude of the Main Interaction Term

(DCA = distance to comparative advantage)



Figure 1: Kaplan-Meier Survival Functions for Liberalizing Countries I

Figure (1) compares Kaplan-Meier survival functions of products belonging to the 25th and 75th percentiles (i.e, to the bottom and the top 25th percentile) of the distance to comparative advantage **before** the liberalization episodes. The more steeply these functions decrease after the initial year 0, the lower is the average survival probability of the related product groups.



Figure 2: Kaplan-Meier Survival Functions for Liberalizing Countries II

Figure (2) compares Kaplan-Meier survival functions of products belonging to the 25th and 75th percentiles (i.e, to the bottom and the top 25th percentile) of the distance to comparative advantage **after** the liberalization episodes. The more steeply these functions decrease after the initial year 0, the lower is the average survival probability of the related product groups.

Spells
- Export
Statistics
riptive 3
: Desc
Table 1

The table provides descriptive statistics for the export spells (continuous exporting of product k from country c to the world market) both for the full sample including also countries where the stock market was closed or open to foreign investors during the whole period of 1980-1997 and for the liberalizing subsample including only countries where the event of stock market liberalization occurred between 1980 and 1997, i.e. only countries that "switched" from being financially closed to being financially open. DCA refers to our measure of "distance to comparative advantage", with lower values (e.e., bottom 25th percentile) corresponding to products well aligned with the comparative advantage of the exporting country and

nower values (e.g., potuoni zour percentile) corresponding to products we higher values (e.g., top 25th percentile) corresponding to products far aw	a augued with the comparative advantage	anuage of the exporting country and of the exporting country.
LIBERALIZING SUBSAMPLE	(1) average length of the spell	(2) median length of the spell
Switchers, non-liberalized Switchers, liberalized	1.68 7.44	1.00 3.00
Switchers, non-liberalized, bottom 25th percentile of DCA Switchers, non-liberalized, top 25th percentile of DCA	1.68 1.66	1.00 1.00
Switchers, liberalized, bottom 25th percentile of DCA Switchers, liberalized, top 25th percentile of DCA	7.78 6.46	4.00 3.00
FULL SAMPLE	(1) average length of the spell	(2) median length of the spell
All countries, non-liberalized All countries, liberalized	3.21 5.33	1.00 2.00
All countries, non-liberalized, bottom 25th percentile of DCA All countries, non-liberalized, top 25th percentile of DCA	3.35 2.87	1.00 1.00
All countries, liberalized, bottom 25th percentile of DCA All countries, liberalized, top 25th percentile of DCA	5.40 4.94	2.00 2.00

Table 2: Summary Statistics

The table provides information on number of observations, mean value, standard deviation, and minimum and maximum values

for all variables in our sample.					
Variable	Obs	Mean	Std. Dev.	Min	Max
StM liberalization	118,542	0.334	0.472	0	1.000
Distance to comparative advantage	118,542	1.388	0.816	0.019	6.577
StM Liberalization x Distance to comparative advantage	118,542	0.473	0.836	0	6.577
Banks x Distance to comparative advantage	107,852	0.442	0.600	0.002	8.187
GDP pc x Distance to comparative advantage	116,375	11.504	7.430	0.160	64.479
Stock Markets x Distance to comparative advantage	50,828	0.292	0.642	0.000	12.424
Banks	107,852	0.290	0.228	0.014	2.179
log GDP pc of exporter in constant PPP dollars	116,375	8.236	0.954	6.166	10.384
Stock Markets	50,828	0.205	0.274	0.001	2.824
StM liberalization x External Finance Dependence	87, 338	0.085	0.179	-0.451	1.140
External Finance Dependence x Banks	79,533	0.075	0.092	-0.695	1.734
External Finance Dependence	87, 338	0.284	0.243	-0.451	1.140
Trade liberalization x Distance to comparative advantage	118,542	0.661	0.879	0	6.577
Trade liberalization	118,542	0.497	0.500	0	1.000
Trade openness x Distance to comparative advantage	118,542	0.948	1.367	0.004	24.563
Trade openness	118,542	0.646	0.374	0.063	4.253
GDP pc	116,375	8.236	0.954	6.166	10.384
Physical capital x Phys. cap. intensity	87, 338	0.700	0.307	0.108	2.345
Human capital x Hum. cap. intensity	87, 338	4.699	2.753	0.196	19.581
Physical capital	118,542	9.448	1.315	5.974	12.073
Human capital	118,542	4.599	2.389	0.366	12.206
Land endowment	118,542	0.652	0.650	0.001	6.550
Number of suppliers	118,542	4.654	0.409	0.693	5.361
Total exports	118,542	14.566	1.658	10.137	20.359
Initial exports	118,542	2.215	1.886	-0.691	13.723
Number of spells	118.542	5.202	2.061	1.000	13.000

 Table 3: Stock Market Liberalization and Long-Term Export Survival - Baseline

product k as strata variable. Our main variable of interest is the interaction between the stock market liberalization dummy and the distance of exported product from the comparative foreign investors to acquire equity stakes in domestic firms before or during the life of a given export spell) and distance to comparative advantage (defined as Euclidean distance of the revealed factor intensity of exported product k to the factor endowment of exporting country c). Product-level control variables include number of suppliers (i.e., number of countries The dependent variable is the hazard rate for an export spell of product k from country c to the world market that started at time t_0 . All regressions are estimated using the Cox Proportional Hazard Model (Cox, 1972) and control for exporting-country and time fixed effects. Estimations also allow for different baseline hazard rate across products by defining advantage of the exporting country. The control variables include also the direct effects of stock market liberalization dummy (which equals one if and only if a given country allowed exporting a given product to the world market) measured at the product-time level, total value of country's exports to the rest of the world (country-time level), initial value of the export of a given product (country-product-time level), number of export spells during the sample period for a given country-product pair (country-product level). The remaining control variables are GDP per capita of exporting country, interactions of physical and human capital at the country level with the corresponding capital intensities at the industry level, and spell to. We report coefficients and not hazard ratios. Robust standard errors clustered at (exporting country)*time (c * to) level are in parentheses. *, **, and *** denote statistical the direct effects of exporting countries' endowments of physical capital, human capital and land. All time-varying explanatory variables are measured in the initial year of the export

significance at the 10% , 5% , and 1% levels, respectively.				
	(1) All	(2) Lib	(3) All	(4)Lib
StM liberalization x Distance to comparative advantage	0.143^{***}	0.152^{***}	0.161^{***}	0.152^{***}
	(0.018)	(0.021)	(0.024)	(0.026)
StM liberalization	-1.116^{***}	-1.655^{***}	-1.184^{***}	-1.697^{***}
	(0.041)	(0.050)	(0.047)	(0.054)
Distance to comparative advantage	0.115^{***}	0.031^{**}	0.082^{***}	0.004
	(0.009)	(0.014)	(0.011)	(0.016)
Number of suppliers	-0.372***	-0.340^{***}	-0.076*	-0.015
	(0.019)	(0.024)	(0.045)	(0.059)
Total exports	0.109^{***}	0.126^{***}	0.008	0.233^{***}
	(0.019)	(0.037)	(0.027)	(0.040)
Initial export	-0.073***	-0.069***	-0.073***	-0.068***
	(0.002)	(0.002)	(0.002)	(0.003)
Number of spells	0.112^{***}	0.108^{***}	0.115^{***}	0.112^{***}
	(0.002)	(0.003)	(0.002)	(0.004)
GDP pc			0.450^{***}	-0.010
			(0.076)	(0.120)
Physical capital x Phys. cap. intensity			-0.080	-0.450^{**}
			(0.103)	(0.179)
Human capital x Hum. cap. intensity			-0.030***	-0.031^{***}
			(0.001)	(0.010)
Physical capital			-0.147^{***}	-0.303^{***}
			(0.054)	(0.096)
Human capital			0.003	-0.063*
			(0.028)	(0.037)
Land endowment			0.135^{*}	0.448^{***}
			(0.071)	(0.134)
Exporting country FE	Yes	Yes	Yes	\mathbf{Yes}
Time FE	\mathbf{Yes}	\mathbf{Yes}	$\mathbf{Y}_{\mathbf{es}}$	\mathbf{Yes}
Product stratification	Yes	Yes	Yes	Yes
Observations	118,542	60,903	85,675	45,699

Table 4: Stock Market Liberalization and Long-Term Export Survival - Alternative Financial Channels
The dependent variable is the hazard rate for an export spell of product k from country c to the world market that started at time t_0 . All regressions
are estimated using the Cox Proportional Hazard Model (Cox, 1972) and control for exporting-country and time fixed effects. Estimations also allow for
different baseline hazard rate across products by defining product k as strata variable. External finance dependence at the industry level is defined as
capital expenditures minus cash flow from operations, divided by capital expenditures, for the median firm in a given industry. Banks is a proxy for the
domestic banking development in the exporting country, measured as the ratio of the bank credit to private sector over the country's GDP. Stock markets
is a proxy for the depth of domestic stock market in the exporting country, measured as the ratio of the stock market capitalization over the country's
GDP. Other variables are defined in Table 3. Full set of controls corresponds to column (4) of Table 3. We report coefficients and not hazard ratios.
Robust standard errors clustered at (exporting country) [*] time $(c * t_0)$ level are in parentheses. *, **, and *** denote statistical significance at the 10%,
5%, and 1% levels, respectively.

0/0, and 1/0 revers, respectively.						
	(1)	(2)	(3)	(4)	(5)	(9)
	All	Lib	All	Lib	All	Lib
StM liberalization x Distance to comparative advantage	0.158^{***}	0.149^{***}	0.130^{***}	0.133^{***}	0.062^{**}	0.060^{**}
	(0.024)	(0.026)	(0.025)	(0.028)	(0.027)	(0.030)
StM liberalization	-1.118^{***}	-1.643^{***}	-1.167^{***}	-1.635^{***}	-1.499^{***}	-1.538^{***}
	(0.048)	(0.054)	(0.049)	(0.058)	(0.058)	(0.062)
Distance to comparative advantage	0.079^{***}	0.005	0.039^{***}	-0.060***	0.050^{**}	0.040^{*}
	(0.011)	(0.016)	(0.015)	(0.023)	(0.020)	(0.021)
StM liberalization x External Finance Dependence	-0.229^{***}	-0.187^{***}				
	(0.036)	(0.042)				
Banks x Distance to comparative advantage			0.121^{***}	0.180^{***}		
			(0.037)	(0.054)		
Banks			0.195^{*}	-0.101		
			(0.111)	(0.144)		
Stock Markets x Distance to comparative advantage					0.020	0.040
					(0.043)	(0.057)
Stock Markets					0.179^{*}	0.128
					(0.102)	(0.114)
Full set of controls	Yes	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$	\mathbf{Yes}	$\mathbf{Y}_{\mathbf{es}}$
Exporting country FE	${ m Yes}$	$\mathbf{Y}_{\mathbf{es}}$	\mathbf{Yes}	\mathbf{Yes}	${ m Yes}$	${ m Yes}$
Time FE	\mathbf{Yes}	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$	\mathbf{Yes}	${ m Yes}$
Product stratification	Yes	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	$\mathbf{Y}_{\mathbf{es}}$
Observations	85,675	45,699	77,986	41,271	37,056	32,892

The dependent variable is the hazard rate for an export the Cox Proportional Hazard Model (Cox, 1972) and cc products by defining product k as strata variable. Trac dummy variable based on trade liberalization dates froi are defined in Table 3. Full set of controls correspond (exporting country)*time $(c * t_0)$ level are in parenthese	spell of produ- ontrol for expoi de openness is m Wacziarg an ls to column (es. *, **, and *	the from country i ting-country i defined as the defined as the d Welch (2008 t) of Table 3.	try c to the we and time fixed sum of counti), who update We report co tistical signific	orld market th effects. Estim. :y's exports an d the previous efficients and ance at the 10	at started at ti ations also allo id imports divi database by S not hazard rat %, 5%, and 1%	me t_0 . All regr w for different ided by its GL bachs and War clos. Robust s for levels, respec	ressions are est baseline hazar DP. Trade liber ner (1995). Ot tandard errors tively.	imated using d rate across alization is a her variables clustered at
	(1) All	(2)Lib	(3) All	(4) Lib	(5) All	(6)Lib	(7) All	(8) Lib
StM liberalization x Distance to comp. adv.	0.172^{***}	0.145^{***}	0.157^{***}	0.146^{***}	0.168^{***}	0.140^{***}	0.127^{***}	0.140^{***}
StM liberalization	(0.027) -1.197***	(0.026) -1.689***	(0.024)-1.179***	(0.026)-1.692***	(0.027) -1.192***	(0.026)-1.685***	(0.025)-1.033***	(0.029) -1.584***
Distance to commenciation educations	(0.051)	(0.054)	(0.047)	(0.054)	(0.050)	(0.054)	(0.051)	(0.056)
Discalle to comparative any antiage	(0.093)	(0.130)	(0.019)	(0.025)	(0.093)	(0.129)	(0.013)	(0.019)
GDP pc x Distance to comp. adv.	-0.020°	0.040^{**}	~	~	-0.020°	0.033^{**}	~	
Trade openness x Distance to comp. adv.			-0.022	0.048	-0.014	0.038		
Trade openness			-0.145^{*}	(1.0.0) -0.288**	-0.155*	(0.00)-0.270**		
			(0.078)	(0.120)	(0.080)	(0.120)		
Trade liberalization x Distance to comp. adv.							0.038^{**} (0.019)	-0.041 (0.030)
Trade liberalization							-0.634^{***}	-0.552^{***}
							(0=0.0)	(100.0)
Full set of controls	\mathbf{Yes}	Yes	\mathbf{Yes}	\mathbf{Yes}	Yes	\mathbf{Yes}	Yes	$\mathbf{Y}_{\mathbf{es}}$
Exporting country FE	\mathbf{Yes}	Yes	Y_{es}	Yes	Yes	Yes	\mathbf{Yes}	${ m Yes}$
Time FE	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}
Product stratification	Yes	Yes	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}
Observations	85,675	45,699	85,675	45,699	85,675	45,699	85,675	45,699

Table 5: Stock Market Liberalization and Long-Term Export Survival - Alternative Non-Financial Channels

Table 6: Stock Market Liberalization and Long-Term Export Survival - Alternative Channels Combined

The dependent variable is the hazard rate for an export spell of product k from country c to the world market that started at time t_0 . All regressions
are estimated using the Cox Proportional Hazard Model (Cox, 1972) and control for exporting-country and time fixed effects. Estimations also allow for
different baseline hazard rate across products by defining product k as strata variable. Variables are defined in Table 3, Table 4, and Table 5. Full set of
controls corresponds to column (4) of Table 3. We report coefficients and not hazard ratios. Robust standard errors clustered at (exporting country)*time
$(c * t_0)$ level are in parentheses. $* **$ and $***$ denote statistical significance at the 10% 5% and 1% levels. respectively.

$(c * t_0)$ level are in parentheses. *, **, and *** denote statistical sign	ificance at the	10%, 5%, and	1% levels, res	pectively.		
	(1)	(2)	(3)	(4)	(2)	(9)
	ĂÌÌ	Lib	ĂIJ	Ĺib	ĂIJ	Ĺib
StM liberalization x Distance to comparative advantage	0.125^{***}	0.137^{***}	0.132^{***}	0.124^{***}	0.052^{*}	0.050^{*}
	(0.025)	(0.029)	(0.027)	(0.028)	(0.028)	(0.030)
StM liberalization	-0.971^{***}	-1.531^{***}	-1.169^{***}	-1.625^{***}	-1.486^{***}	-1.524^{***}
	(0.051)	(0.057)	(0.051)	(0.058)	(0.059)	(0.062)
Distance to comparative advantage	0.059^{***}	0.015	0.172^{*}	-0.334^{**}	-0.115	-0.087
	(0.013)	(0.019)	(0.094)	(0.140)	(0.143)	(0.164)
StM liberalization x External Finance Dependence	-0.218^{***}	-0.180^{***}				
	(0.036)	(0.042)				
Trade liberalization x Distance to comparative advantage	0.038^{**}	-0.039				
	(0.019)	(0.030)				
Trade liberalization	-0.634^{***} (0.043)	-0.553^{***} (0.054)				
Banks x Distance to comparative advantage			0.143^{***}	0.137^{**}		
			(0.040)	(0.064)		
Banks			0.163	0.009		
			(0.118)	(0.157)		
Stock Markets x Distance to comparative advantage					-0.002	0.054
Stock Markets					(0.047)	(0.000) 0.205
					(0.122)	(0.131)
GDP pc x Distance to comparative advantage			-0.014	0.034^{**}	0.020	0.018
			(0.012)	(0.017)	(0.017)	(0.020)
Trade openness x Distance to comparative advantage			-0.035	0.005	0.007	-0.040
			(0.029)	(0.042)	(0.041)	(0.048)
Trade openness			-0.076	-0.170	-0.311^{*}	-0.232 (0 166)
			(100.0)	(001.0)	(601.0)	(001.0)
Full set of controls	Yes	Yes	Yes	Yes	\mathbf{Yes}	Yes
Exporting country FE	Yes	Yes	Yes	Yes	Yes	Yes
Time FE	Yes	Yes	Yes	Yes	Yes	Yes
Product stratification	Yes	Yes	Yes	Yes	Yes	Yes
Observations	85,675	45,699	77.986	41.271	37,056	32,892

Table 7: Robustness Checks I - Additional Controls

The dependent variable is the hazard rate for an export spell of product k from country c to the world market that started at time t_0 . All Estimations also allow for different baseline hazard rate across products by defining product k as strata variable. The real exchange rate is or price-referenced in trade publications are from Rauch (1999). The IMF dummy captures whether a given country was a subject to the Other variables are defined in Table 3. Full set of controls corresponds to column (4) of Table 3. We report coefficients and not hazard regressions are estimated using the Cox Proportional Hazard Model (Cox, 1972) and control for exporting-country and time fixed effects. taken from the World Development Indicators database. The data identifying products as homogenous (omitted category), differentiated, ratios. Robust standard errors clustered at (exporting country)^{*}time ($c * t_0$) level are in parentheses. *, **, and *** denote statistical IMF program (i.e., received an official IMF loan) in a given year and is constructed based on data from the International Monetary Fund. significance at the 10%. 5% and 1% levels re

aignincance at the 10%, 3%, and 1% levels, respectively.					
	(1) Lib	(2)	(3) Lih	(4) Lib	(5) Lib
StM liberalization x Distance to comparative advantage	0.218^{***}	0.152^{***}	0.126^{***}	0.237^{***}	0.218^{***}
	(0.031)	(0.027)	(0.026)	(0.035)	(0.037)
StM liberalization	-1.760^{***}	-1.556^{***}	-1.610^{***}	-1.794^{***}	-1.815^{***}
	(0.069)	(0.062)	(0.055)	(0.063)	(0.065)
Distance to comparative advantage	-0.010	0.007	0.009	0.171^{***}	0.149^{***}
	(0.050)	(0.017)	(0.021)	(0.047)	(0.046)
Real exchange rate x Distance to comparative advantage	0.009				
	(0.025)				
Real exchange rate	0.104^{***}				
	(0.033)				
StM liberalization x Differentiated products		-0.245^{***}			
		0000)			
5tM liberalization x Price-referenced products		-0.041 (0.033)			
IMF dummy x Distance to comparative advantage			0.015		
			(0.033)		
IMF dummy			-0.461***		
			(ccu.u)		
Time trend x Distance to comparative advantage				-0.011^{***}	
Exporter-time trend x Distance to comparative advantage Exporter-time trend					${ m Yes}{ m Yes}$
interface of controls	Voc	Vec	Vac	Vac	Vac
	100		100	100	100
Exporting country FE	Yes	Yes	Yes	Yes	Yes
Lime FE	Yes	Yes	Yes	Yes	Yes
Product stratification	Yes	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	Yes
Observations	$23,\!829$	$42,\!634$	45,699	45,699	45,699

ixed Effects
Ē
~
, and
cation.
Ч
÷Ę
Ľa
£
0
60
rin
E
Cluste
- Cluste
i II - Cluste
thecks II - Cluste
Checks II - Cluste
oustness Checks II - Cluste
obustness Checks II - Cluste
Robustness Checks II - Cluste
Robustness Checks II - Cluste
8: Robustness Checks II - Cluste
e 8: Robustness Checks II - Cluste
ole 8: Robustness Checks II - Cluste
able 8: Robustness Checks II - Cluste

The dependent variable is the hazard rate for an export spell of product k from country c to the world market that started at time t_0 . All regressions are estimated using the Cox Proportional Hazard Model (Cox, 1972). Variables are defined in Table 3. Regressions in columns (1)-(3) control for exporting-country and time fixed effects, and also allow for different baseline hazard rate across products by defining product k as strata variable. In column (4), we replace product stratification by the more stringent product*time stratification. In column (5), we replace product stratification by the more stringent product*(exporting-country) stratification. In column (6), we replace the separate exporting-country and time fixed effects by the interacted (exporting-country)*time fixed effects. Column (7) presents results without any fixed effects. Full set of controls corresponds to column (4) of Table 3. We report coefficients and not hazard ratios. Robust standard errors are in parentheses. They are clustered at (exporting country)^{*}time ($c * t_0$) level in columns (1) and (4)-(7). In the second column, they are clustered at the exporting country level. In the third column, we follow the two-way procedure suggested by Cameron, Gelbach, and Miller (2006) and cluster the standard errors alongside both country and time dimension. ماميسا مر امسار - +1- 1007 E07 oto stotististio * * *

γ , γ , and γ , ord for the statistical significance at the 10%, 5%, and 1%	o levels, respect	uvely.					
	(1) Lib	(2)Lib	(3) Lib	(4) Lib	(5)Lib	(6)Lib	(7) Lib
StM Liberalization x Distance to comparative advantage	0.152^{***}	0.152^{***}	0.152^{***}	0.146^{***}	0.180^{***}	0.168^{***}	0.190^{***}
	(0.026)	(0.054)	(0.054)	(0.028)	(0.029)	(0.028)	(0.035)
StM liberalization	-1.697^{***}	-1.697^{***}	-1.697^{***}	-1.734^{***}	-1.351^{***}	-2.296^{***}	-1.130^{***}
	(0.054)	(0.087)	(0.150)	(0.064)	(0.055)	(0.059)	(0.065)
Distance to comparative advantage	0.004	0.004	0.004	0.031	-0.093***	0.001	-0.028^{*}
	(0.016)	(0.026)	(0.026)	(0.020)	(0.029)	(0.016)	(0.015)
Full set of controls	Yes	Yes	Yes	Yes	Yes	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$
Exporting country FE	\mathbf{Yes}	${ m Yes}$	${ m Yes}$	\mathbf{Yes}			
Time FE	Yes	\mathbf{Yes}	\mathbf{Yes}		\mathbf{Yes}		
(Exporting-country)*Time FE						\mathbf{Yes}	
Product stratification	Y_{es}	\mathbf{Yes}	${ m Yes}$			\mathbf{Yes}	
Product*Time stratification				\mathbf{Yes}			
Product*(Exporting-country) stratification					Yes		
Observations	45.699	45.699	45.699	45.699	45.699	45.699	45.699

	2	
	Ľ	
	0	
	≥	
	ω	
	P	
	Я	
	ч	
ſ	T.	
_	_	
	ğ	
	2	
Ī	5	
	Ч	
	Ξ	
C	J	
	υ	
-	9	
	÷	
	d	
•	Ξ	
	S	
	Õ	
	Ξ	
	30	
F	-	
	ы	•
	ã	~
•	Ξ	
	\square	
	g	
÷	Ξ	
Ė	E F	
Ė	- Tin	
Ē	011. T T	
Ē	01.T TT	
Ē		
E FFF	cs III - Tin	
	CKS III - TID	
	ecks III - Tin	
	hecks III - Tin	
	Unecks III - Tin	
	Checks III - Tin	
	ss Checks III - Tin	
	ess Checks III - Tin	
	ness Checks III - Tin	
	stness Checks III - Tin	
	ustness Checks III - Tin	
	oustness Checks III - Tin	
	obustness Checks III - Tin	
	Cobustness Checks III - Tin	
	Kobustness Checks III - Tin	
	1: Kobustness Checks III - Tin	
	9: Kobustness Checks III - Tin	
	e 9: Kobustness Checks III - Tin	
	ole 9: Kobustness Checks III - Tin	
	able 9: Kobustness Checks III - Tin	
	Lable 9: Kobustness Checks III - Tin	

The dependent variable is the hazard rate for an export spell of product k from country c to the world market that started at time t_0 . All regressions are estimated using the Cox Proportional Hazard Model (Cox, 1972) and control for exporting-country and time fixed effects. Estimations also allow for different baseline hazard rate across products by defining product k as strata variable. Variables are defined in Table 3. Column (1) presents results when Distance to comparative advantage) to vary over the life of the spells. Column (2) presents the results for the survival database constructed using exports data between 1980-1997. Column (3) shows the results of the estimation on a restricted sample that contains spells whose full length occurs after or we allow the effects of our main interaction term and its two components (StM liberalization x Distance to comparative advantage, StM liberalization, before the liberalization episodes (i.e., dropping "partially treated" spells). In column (4), we use the Efron method (Efron, 1988) to handle tied-spells he clustered ands to column (4) of Table 3. We remort coefficients and not hazard ratios. Bohust standard emtermination events Full set of controls of

bermination events. Furtiset of controls corresponds to commut (4) of at (exporting-country) [*] time $(c * t_0)$ level are in parentheses. *, **,	and *** denote	port coenicien e statistical sig	es and not naza gnificance at th	e 10%, 5%, and 1% lev	aru errors clustereu els, respectively.
	(1)		(2)	(3)	(4)
	tvc	tvc	1980 - 1997	dropping spells	tied-spells
StM liberalization x Distance to comparative advantage	0.120^{***}	0.018	0.160^{***}	0.110^{***}	0.171^{***}
	(0.042)	(0.014)	(0.030)	(0.021)	(0.030)
StM liberalization	-0.717***	-0.299***	-1.810^{***}	-0.973***	-2.058***
	(0.069)	(0.019)	(0.061)	(0.042)	(0.060)
Distance to comparative advantage	0.013	-0.005	0.010	0.008	0.012
	(0.025)	(0.012)	(0.016)	(0.013)	(0.020)
Full set of controls	Yes	I	$\mathbf{Y}_{\mathbf{es}}$	m Yes	${ m Yes}$
Exporting country FE		I	$\mathbf{Y}_{\mathbf{es}}$	Yes	Yes
Time FE		I	$\mathbf{Y}_{\mathbf{es}}$	Y_{es}	\mathbf{Yes}
Product stratification	Yes	I	Yes	${ m Yes}$	${ m Yes}$
Observations	45.699	I	45.699	33.110	45.699

The dependent variable is the hazard rate for an export spell of pro- Cox Proportional Hazard Model (Cox, 1972) and control for exporti- by defining product k as strata variable. Variables are defined in Tab that experienced stock market liberalization during the 1980-1997 _I System and Property Rights". Columns (2) and (5) report results for with above-median value for this legal index. Full set of controls co	duct k from col ng-country and ole 3. Columns period. Column or countries wit	untry c to the worl I time fixed effects. (1)-(3) show result as (1) and (4) repc th below-median ve blumn (4) of Table	d market that star Estimations also a s for the whole sam rrt results for all co lue for this legal ir 3. We report coeff	ted at time t_0 . Jallow for different ple and columns ountries with av out columns (dex. Columns (ficients and not	All regressions are e t baseline hazard ra (4)-(6) for the subs ailable data on the 3) and (6) report re hazard ratios. Robu	stimated using the te across products ample of countries legal index "Legal sults for countries is standard errors
clustered at (exporting country). time $(c * t_0)$ level are in parentneses	s. , , and	denote statistica.	signincance at the	10%, 5%, and 1	70 levels, respective	ly.
	(1)	(2)	(3)	(4)	(5)	(9)
	Áll	Åll	Áll	Ĺib	Ĺib	Ĺib
	legal index	legal index	legal index	legal index	legal index	legal index
	available	below median	above median	available	below median	above median
StM liberalization x Distance to comparative advantage	0.149^{***}	0.070^{*}	0.246^{***}	0.138^{***}	0.022	0.275^{***}
	(0.025)	(0.036)	(0.035)	(0.027)	(0.039)	(0.033)
StM liberalization	-1.187^{***}	-1.058^{***}	-1.446^{***}	-1.694^{***}	-1.602^{***}	-1.847***
	(0.049)	(0.064)	(0.072)	(0.055)	(0.071)	(0.080)
Distance to comparative advantage	0.074^{***}	0.095^{***}	-0.028	0.007	0.065^{***}	-0.162^{***}
	(0.013)	(0.015)	(0.028)	(0.016)	(0.021)	(0.032)
Full set of controls	\mathbf{Yes}	\mathbf{Yes}	Yes	Yes	\mathbf{Yes}	\mathbf{Yes}
Exporting country FE	\mathbf{Yes}	\mathbf{Yes}	${ m Yes}$	Yes	\mathbf{Yes}	Yes
Time FE	${ m Yes}$	\mathbf{Yes}	${ m Yes}$	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}
Product stratification	Yes	\mathbf{Yes}	\mathbf{Yes}	Yes	$\mathbf{Y}_{\mathbf{es}}$	\mathbf{Yes}
Observations	77,165	54,154	23,011	42,376	26, 126	16,250

Table 10: Stock Market Liberalization and Long-Term Export Survival, Below and Above of the Median of Legal System

54

Stock Market Liberalizations and Export Dynamics Online Appendix

 $\label{eq:Madina Kukenova^{\dagger}} Martin \ Strieborny^{\ddagger}$

This version: June 5, 2019

^{*}World Bank; E-Mail: mjaud@worldbank.org

[†]Webster University Geneva; E-Mail: kukenova@webster.ch

[‡]Knut Wicksell Centre for Financial Studies, Lund University, P.O. Box 7082, S-220 07 Lund, Sweden. E-Mail: Martin.Strieborny@nek.lu.se

D Alternative Measures of Stock Market Liberalization: Tables I-IX

Tables I to IX re-run the estimations from Tables 3-7 in the main text, replacing the official liberalization dummy by two alternative measures of stock market liberalization - first sign liberalization dummy and a continuous variable measuring the stock market liberalization intensity. Table I corresponds to Table 3 in the main text, Table II and Table III correspond to Table 4 in the main text, Table IV and Table V correspond to Table 5 in the main text, Table VI and Table VII correspond to Table 6 in the main text, and Table VIII and Table IX correspond to Table 7 in the main text.

E Alternative Measures of Legal Environment: Tables X-XIV

Tables X to XIV re-run the estimations form Table 10 in the main text, replacing the sample split based on the variable Legal System and Property Rights by sample splits based on Efficiency of Judicial System (an assessment of the "efficiency and integrity of the legal environment as it affects business, particularly foreign firms" produced by the country-risk rating agency Business International Corp.) in Table X, Rule of Law (an assessment of the law and order tradition in the country produced by the country-risk rating agency International Country Risk (ICR)) in Table XI, Risk of Expatriation (an ICR's assessment of the risk of "outright confiscation" or "forced nationalization") in Table XII, Repudiation of Contracts by Government (an ICR's assessment of the "risk of modification in a contract taking the form of repudiation, postponement or scaling down" due to "budget cut-back, indigenization pressure, a change in government, or change in government economic and social priorities.") in Table XIII, and Investor Protection (average value of Rule of Law, Risk of Expropriation, and Repudiation of Contracts by Government) in Table XIV.

VARIABLES	(1) All	(2) Lib	(3) All	(4) Lib	(5) All	(6) Lib	(7) All	(8) Lib
StM liberalization intensity x Distance to comparative advantage	0.283^{***}	0.628***	0.393***	0.754***				
StM liberalization intensity	(0.022) -2.292***	(0.055) -2.935***	(0.036) -2.548***	(0.071) -3.234***				
StM liberalization first sign x Distance to comparative advantage	(060.0)	(0.142)	(0.1.00)	(691.0)	0.101^{***}	0.158^{***}	0.126^{***}	0.152^{***}
StM liberalization first sign					(0.011) -0.994***	(0.020) -1.457***	(0.017) -1.063***	(0.024) -1.498***
Distance to comparative advantage	0.112^{***}	0.050***	0.084^{***}	0.032^{**}	(0.030) 0.114^{***}	(0.048) 0.024^{*}	(0.040) 0.078^{***}	(1 c 0.0) -0.004
Number of suppliers	(0.008) -0.399***	(0.014) - $0.376***$	(0.010)	(0.015) 0.038	(0.008) -0.374***	(0.013) - 0.340^{***}	(0.011) -0.094**	(0.016) -0.051
Total exports	(0.020) 0.055^{***}	(0.027) 0.232^{***}	(0.046) 0.044^{*}	(0.063) 0.311^{***}	(0.019) 0.084^{***}	(0.023) 0.077**	(0.046) 0.003	(0.061) 0.211^{***}
Initial export	(0.017) -0.076***	(0.037) - $0.078***$	(0.025) -0.077***	(0.053) -0.078***	(0.018) - 0.074^{***}	(0.036) -0.072***	(0.025) - 0.074^{***}	(0.041) - 0.071^{***}
Number of spells	(0.002) 0.122^{***}	(0.003) 0.137^{***}	(0.002) 0.126^{***}	(0.003) 0.142^{***}	(0.002) (0.115^{***})	(0.002) 0.116^{***}	(0.002) (0.118^{***})	(0.003) 0.121^{***}
GDP pc	(200.0)	(0.003)	(0.002) -0.072 (0.069)	(0.003) -0.543*** (0.115)	(0.002)	(600.0)	(0.371^{***})	(0.004) -0.002 (0.115)
Physical capital x Phys. cap. intensity			(0.002)	-0.361** -0.361**			(270.0) -0.078 (201.0)	-0.452** -0.452**
Human capital x Hum. cap. intensity			-0.031^{***}	-0.032***			-0.028***	-0.030***
Physical capital			(0.007) 0.131*** (0.048)	0.235*** 0.235***			-0.154*** -0.154***	-0.377*** -0.377***
Human capital			(0.040) -0.032 (0.033)	-0.114*** -0.114***			-0.003 -0.003	-0.045 -0.045
Land endowment			(0.060) (0.060) (0.061)	(0.130)			$\begin{array}{c} (0.020) \\ 0.180^{***} \\ (0.065) \end{array}$	(660.0) 0.669^{***} (0.119)
Exporting country FE Time FF	Y_{es}	Yes Ves	Yes Ves	Yes Ves	Yes Ves	Yes	$\mathbf{Y}_{\mathbf{PS}}$	Yes Ves
Product stratification	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Observations	118,542	60,903	85,675	45,699	118,542	60,903	85,675	45,699

Table I: Other Proxies of Stock Market Liberalization - Baseline

VARIABLES	(1) All	$^{(2)}$ Lib	(3) All	(4) Lib	(5) All	(6) Lib
StM liberalization intensity x Distance to comparative advantage	0.390***	0.748^{***}	0.350^{***}	0.721^{***}	0.293^{***}	0.517^{***}
-	(0.036)	(0.071)	(0.037)	(0.071)	(0.050)	(0.080)
StM liberalization intensity	-2.452^{***}	-3.080***	-2.491^{***}	-3.243^{***}	-3.266***	-3.526***
	(0.108)	(0.165)	(0.111)	(0.171)	(0.160)	(0.194)
Distance to comparative advantage	0.083^{***}	0.032^{**}	0.043^{***}	-0.060***	0.037^{**}	0.026
	(0.010)	(0.015)	(0.014)	(0.023)	(0.017)	(0.020)
StM liberalization intensity x External Finance Dependence	-0.351^{***} (0.085)	-0.572^{***} (0.132)				
Banks x Distance to comparative advantage	~	~	0.111^{**}	0.220^{***}		
			(0.044)	(0.052)		
Banks			-0.048	-0.340^{***}		
			(0.110)	(0.132)		
Stock Markets x Distance to comparative advantage					0.064	0.031
					(0.039)	(0.054)
Stock Markets					0.072	0.107
					(0.093)	(0.097)
Full set of controls	Yes	Yes	Yes	Yes	Yes	Yes
Exporting country FE	\mathbf{Yes}	$\mathbf{Y}_{\mathbf{es}}$	Yes	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$	\mathbf{Yes}
Time FE	\mathbf{Yes}	$\mathbf{Y}_{\mathbf{es}}$	Yes	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$	\mathbf{Yes}
Product stratification	Yes	Yes	Yes	Yes	Yes	Yes
Observations	85,675	45,699	77,986	41,271	37,056	32,892

Table II: Stock Market Liberalization Intensity - Alternative Financial Channels

VARIABLES	$^{(1)}_{ m All}$	(2) Lib	(3) All	$^{(4)}$ Lib	(5) All	(6) Lib
StM liberalization first sign x Distance to comparative advantage	0.125^{***}	0.149^{***}	0.109^{***}	0.145^{***}	0.071***	0.087***
StM liberalization first sign	(0.017) -1.006***	(0.024)-1.437***	(0.017)-1.045***	(0.026)-1.438***	(0.022)-1.296***	(0.029)-1.336***
Distance to comparative advantage	(0.041) 0.076^{***}	(0.052) -0.002	(0.042) 0.040^{***}	(0.055) -0.058**	(0.053) 0.037^{*}	(0.058) 0.026
StM liheralization first sion x External Finance Denendence	(0.011) -0.197***	(0.016) -0.211***	(0.014)	(0.023)	(0.019)	(0.022)
	(0.031)	(0.040)				
Banks x Distance to comparative advantage			0.106^{***} (0.037)	0.146^{***} (0.053)		
Banks			0.073	(0.140)		
Stock Markets x Distance to comparative advantage					0.023	0.011
					(0.042)	(0.060)
Stock Markets					-0.007	-0.030
					(960.0)	(0.107)
Full set of controls	\mathbf{Yes}	Yes	Yes	Yes	Yes	Yes
Exporting country FE	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$	Yes	Yes	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$
Time FE	$\mathbf{Y}_{\mathbf{es}}$	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	$\mathbf{Y}_{\mathbf{es}}$
Product stratification	Yes	Yes	Yes	Yes	Yes	Yes
Observations	85,675	45,699	77,986	41,271	37,056	32,892

Table III: Stock Market Liberalization First Sign - Alternative Financial Channels

VARIABLES	$^{(1)}_{\rm All}$	(2) Lib	(3) All	(4) Lib	(5) All	(6) Lib	(7) All	(8) Lib
StM liberalization intensity x Distance to comparative advantage	0.469^{***}	0.775***	0.389***	0.748***	0.463^{***}	0.779***	0.301^{***}	0.608***
StM liberalization intensity	(0.044) -2.653***	(0.077) -3.261*** (0.170)	(0.036) -2.546***	(0.071) -3.232***	(0.043) -2.647*** (0.114)	(0.076) -3.272***	(0.038) -2.120*** (0.199)	(0.080) -2.702***
Distance to comparative advantage	(0.115) 0.493^{***}	(0.170) 0.227 (0.140)	(0.111 0.111*** 0.010	(0.104) (0.005)	(0.114) 0.504^{***}	(0.109) 0.278^{*}	(0.122) 0.051^{***}	(0.194) 0.058^{***}
GDP pc x Distance to comparative advantage	-0.051^{***}	(0.140) -0.023	(QTN'N)	(070.0)	-0.050^{***}	-0.034^{*}	(710.0)	(020.0)
Trade openness x Distance to comparative advantage	(210.0)	(810.0)	-0.039	0.049	(0.012)	0.061^{*}		
Trade openness			(0.024) -0.152**	(0.030) -0.308*** (0.110)	$(0.024) - 0.174^{***}$	(0.036) -0.328***		
Trade liberalization \mathbf{x} Distance to comparative advantage			(con.u)	(711.0)	(000.0)	(ett.0)	0.064^{***}	-0.061^{**}
Trade liberalization							(0.018) -0.752*** (0.038)	(0.028) -0.707*** (0.058)
Full set of controls	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Exporting country FE	${ m Yes}_{{ m Voc}}$	Yes	Yes	${ m Yes}_{ m Voc}$	${ m Yes}_{ m voc}$	${ m Yes}_{ m Voc}$	${ m Yes}_{ m Voc}$	${ m Yes}_{ m Voc}$
Product stratification	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Observations	85,675	45,699	85,675	45,699	85,675	45,699	85,675	45,699

Table IV: Stock Market Liberalization Intensity - Alternative Non-Financial Channels

VARIABLES	(1) All	(2) Lib	(3) All	(4) Lib	(5) All	(6) Lib	(7) All	(8) Lib
StM liberalization first sign x Distance to comparative advantage	0.145^{***}	0.144^{***}	0.124***	0.147***	0.143^{***}	0.141***	0.105***	0.158^{***}
StM liberalization first sign	-1.085***	(0.020) -1.489***	-1.060*** -1.060***	(0.024) -1.494***	(0.020) -1.082***	(0.025) -1.487***	() TO () () 17***	-1.383***
Distance to comparative advantage	(0.044) 0.320^{***}	(0.192) -0.297**	(0.040) 0.092^{***}	(100.0) -0.045*	(0.043) 0.329^{***}	(0.052) -0.261**	(0.044) 0.057^{***}	(0.05())
GDP pc x Distance to comparative advantage	-0.030**	(0.135^{**})	(6T0.0)	(070.0)	-0.030^{**}	(0.133) 0.027 (0.016)	(710.0)	(otn.u)
Trade openness x Distance to comparative advantage	(210.0)	(010.0)	-0.019	0.073**	(0.009 -0.009	0.065* 0.065*		
Trade openness			(0.026) -0.142*	(0.037) -0.295** (0.190)	(0.026) -0.156**	(0.037) -0.279** (0.161)		
Trade liberalization \mathbf{x} Distance to comparative advantage			(e10.0)	(071.0)	(670.0)	(171.0)	0.034^{*}	-0.072^{**}
Trade liberalization							(0.018) -0.645*** (0.042)	(0.029) - 0.545^{***} (0.056)
Full set of controls	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Exporting country FE Time FF	Yes Ves	$\mathbf{Y}_{\mathbf{es}}^{\mathbf{es}}$	Yes Ves	$_{\rm Ves}^{\rm Yes}$	$_{\rm Ves}^{\rm Yes}$	$_{\rm Ves}^{\rm Yes}$	Yes Ves	Yes Ves
Product stratification	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Observations	85,675	45,699	85,675	45,699	85,675	45,699	85,675	45,699

Table V: Stock Market Liberalization First Sign - Alternative Non-Financial Channels

	•					
VARIABLES	(1) All	$^{(2)}$ Lib	(3) All	(4) Lib	(5) All	(6) Lib
StM liberalization intensity x Distance to comparative advantage	0.298^{***}	0.598^{***}	0.425^{***}	0.744^{***}	0.354^{***}	0.530^{***}
StM liberalization intensity	-2.021^{***}	-2.527***	-2.594^{***}	-3.273***	-3.349^{***}	-3.556***
Distance to commanative advantage	(0.125) 0.049***	(0.195)	(0.118) 0.558 $***$	(0.180)	(0.172) 0.503***	(0.203) 0.431**
	(0.012)	(0.020)	(0.101)	(0.156)	(0.174)	(0.187)
5tM liberalization intensity x External Finance Dependence	-0.356^{++}	$-0.628^{\pm\pm\pm}$ (0.137)				
Trade liberalization x Distance to comparative advantage	0.064^{***}	-0.058**				
Trade liberalization	-0.753*** -0.753***	-0.711^{***}				
Banks x Distance to comparative advantage	(0000)	(000.0)	0.163^{***}	0.253^{***}		
Banks			(0.045) -0.151	(0.003) -0.373**		
Stock Markets x Distance to comparative advantage			(0.114)	(0.146)	0.074^{*}	0.063
Stock Markets					(0.042) 0.240^{**}	(0.060) 0.268^{**}
					(0.105)	(0.108)
GDP pc x Distance to comparative advantage			-0.063***	-0.044^{**}	-0.052^{**}	-0.044**
Trade openness x Distance to comparative advantage			(etu.u)	0.005	(170.0)	(670.0) -0.067
Trada anamace			(0.026)	(0.042)	(0.039)	(0.046)
			(0.066)	(0.118)	(0.154)	(0.161)
Full set of controls	Yes	Yes	Yes	Yes	Yes	Yes
Exporting country FE	Yes	Yes	Yes	Yes	Yes	${ m Yes}$
Time F'E	Yes	Yes	Yes	Yes	Yes	Yes
Product stratification	Yes	Yes	Yes	Yes	Yes	Yes
Observations	85,675	45,699	77,986	41,271	37,056	32,892

Table VI: Stock Market Liberalization Intensity - Alternative Channels Combined

	0					
VARIABLES	(1) All	(2) Lib	(3) All	(4) Lib	(5) All	(6) Lib
StM liberalization first sign x Distance to comparative advantage	0.103^{***} (0.017)	0.154^{***}	0.121^{***}	0.142^{***} (0.027)	0.066^{***}	0.081^{***} (0.030)
StM liberalization first sign	-0.862***	-1.322***	-1.059***	-1.435***	-1.288***	-1.327***
Distance to comparative advantage	(0.054^{***})	0.015	(0.044) 0.271^{***}	-0.302^{**}	-0.035	-0.100
StM liberalization first sign x External Finance Dependence	(0.012) -0.189***	(0.019) -0.210***	(660.0)	(0.143)	(0.151)	(0.167)
Trade liberalization \mathbf{x} Distance to comparative advantage	0.034* 0.034*	(0.071** -0.071**				
Trade liberalization	-0.644*** 0.644*** 0.041)	$(0.029) - 0.546^{**}$				
Banks x Distance to comparative advantage			0.133^{***}	0.082		
Banks			(0.026 0.026 0.119)	-0.159 -0.159 (0.154)		
Stock Markets x Distance to comparative advantage			(711.0)	(#01.0)	0.008	0.002
Stock Markets					(0.044)	0.044
GDP nc x Distance to comparative advantage			-0.028**	0.027	(0.109)	(0.119) 0.016
			(0.013)	(0.018)	(0.018)	(0.020)
Irade openness x Distance to comparative advantage			-0.024 (0.027)	(0.058)	(0.039)	-0.000
Trade openness			-0.076	-0.150	-0.242	-0.189
			(0.076)	(0.134)	(0.154)	(0.158)
Full set of controls	Yes	Yes	Yes	Yes	Yes	Yes
Exporting country FE	${ m Yes}$	Yes	Yes	Yes	Yes	Yes
1 ime F'E Product stratification	Yes Yes	Yes Yes	Yes Yes	Yes Yes	Yes Yes	Yes Yes
Observations	85,675	45,699	77,986	41,271	37,056	32,892

Table VII: Stock Market Liberalization First Sign - Alternative Channels Combined

	(1)	(2)	(3)	(4)	(2)
VARIABLES	Lib	Lib	Lib	Lib	Lib
StM liberalization intensity x Distance to comparative advantage	0.851^{***}	0.783***	0.662^{***}	0.755***	1.283^{***}
StM liberalization intensity	(0.082) -3.584**	(0.074) -3.111***	(0.072) -2.944***	(0.074) -3.235***	(0.127) -5.131***
Distance to comparative advantage	(0.221) 0.048	(0.192) 0.032^{**}	(0.167) 0.022	(0.167) 0.033	(0.287) 0.201^{***}
Real exchange rate x Distance to comparative advantage	(0.073) 0.018	(010.0)	(170.0)	(160.0)	(0.044)
Real exchange rate	(0.034) 0.076^{**} (0.035)				
StM liberalization intensity x Differentiated products	~	-0.418^{***}			
StM liberalization intensity x Price-referenced products		(0.090) 0.050 (0.001)			
IMF dummy \mathbf{x} Distance to comparative advantage		(TEN'N)	0.046		
IMF dummy			(0.030) -0.569***		
Time trend x Distance to comparative advantage			(000.0)	000.0-	
Exporter-time trend x Distance to comparative advantage Exporter-time trend				(000.0)	$_{\rm Yes}^{\rm Yes}$
Full set of controls	Yes	Yes	Yes	Yes	Yes
Exporting country FE Time FF	Yes Ves	Yes Ves	Yes Ves	${ m Y}_{ m es}$	Yes Ves
Product stratification	Yes	Yes	Yes	Yes	Yes
Observations	23,829	42,634	45,699	45,699	45,699

Table VIII: Stock Market Liberalization Intensity - Additional Controls

VARIABLES	(1) Lib	(2) Lib	(3) Lib	(4) Lib	(5) Lib
StM liberalization first sign x Distance to comparative advantage	0.196^{***}	0.152^{***}	0.128^{***}	0.252^{***}	0.250^{***}
StM liberalization first sign	(0.031) -1.506***	(0.020) -1.310***	(0.020)	(0.032) -1.612^{***}	(0.033) -1.680***
Distance to comparative advantage	(0.068) -0.029	(0.058) -0.001	(0.053) 0.012	(0.058) 0.195^{***}	(0.059) 0.141^{***}
Real exchange rate x Distance to comparative advantage	(0.050) 0.002 (0.002)	(210.0)	(770.0)	(0.040)	(0.042)
Real exchange rate	(020.0) 0.110***				
StM liberalization first sign x Differentiated products	(0.037)	-0.295***			
StM liberalization first sign x Price-referenced products		(0.033) -0.074**			
IMF dummy x Distance to comparative advantage		(0.032)	-0.006 (050.0)		
IMF dummy			-0.459***		
Time trend \mathbf{x} Distance to comparative advantage			(400.0)	-0.014^{***}	
Exporter-time trend x Distance to comparative advantage Exporter-time trend				(enn.n)	Yes Yes
Full set of controls Economics controls	Yes Voc	Yes	Yes	Yes	${ m Yes}_{ m Voc}$
Lapon ung county r L Time FE Denoluor Autorification	Yes	Yes	Yes	Yes	Yes
r routed su additionion Observations	165 73 890	10 637	165 45 600	165 45,600	1 ES 1 F 600
Observations	62,023	42,004	40,039	40,039	40,033

Table IX: Stock Market Liberalization First Sign - Additional Controls

\sim
0
Ц
6
· 🖻
- Ξ
田
r÷i
<u> </u>
a
·8
.≍
5
Ē
Ė.
F
\cup
U
3
• –
D.
ιe.
r-1
θ
É,
تہ
ىب
0
-
9
2
õ
<u>, 0</u>
∕_
٦
5
ы
Ы
⊳
7
1
Θ
\mathbf{m}
Ĵ.
പ്
5
٠÷
2
ы
5
$c \mathbf{n}$
rt '
ort ;
port ;
xport ;
Export ;
Export ;
a Export ;
m Export ;
arm Export ;
lerm Export :
Term Export :
g-Term Export ;
ng-Term Export ;
ong-Term Export ;
ong-Term Export ;
Long-Term Export ;
l Long-Term Export ;
nd Long-Term Export ;
und Long-Term Export ;
and Long-Term Export ;
n and Long-Term Export ;
on and Long-Term Export ?
ion and Long-Term Export ?
vtion and Long-Term Export ?
ation and Long-Term Export ?
ization and Long-Term Export ?
dization and Long-Term Export ?
calization and Long-Term Export ?
sralization and Long-Term Export ?
peralization and Long-Term Export ?
iberalization and Long-Term Export ?
Liberalization and Long-Term Export ?
Liberalization and Long-Term Export ?
Liberalization and Long-Term Export ?
tet Liberalization and Long-Term Export ?
ket Liberalization and Long-Term Export ?
arket Liberalization and Long-Term Export ?
[arket Liberalization and Long-Term Export !
Market Liberalization and Long-Term Export ?
Market Liberalization and Long-Term Export ?
k Market Liberalization and Long-Term Export ?
ck Market Liberalization and Long-Term Export ?
ock Market Liberalization and Long-Term Export ?
tock Market Liberalization and Long-Term Export ?
Stock Market Liberalization and Long-Term Export ?
: Stock Market Liberalization and Long-Term Export ?
X: Stock Market Liberalization and Long-Term Export ?
X: Stock Market Liberalization and Long-Term Export ?
e X: Stock Market Liberalization and Long-Term Export ?
ble X: Stock Market Liberalization and Long-Term Export ?
ble X: Stock Market Liberalization and Long-Term Export ?
Sable X: Stock Market Liberalization and Long-Term Export ?

products by defining product k as strata variable. Variables are defined in Table 3 in the main text. Columns (1)-(3) show results for the whole sample and columns report results for countries with above-median value for this legal index. Full set of controls corresponds to column (4) of Table 3 in the main text. We report coefficients The dependent variable is the hazard rate for an export spell of product k from country c to the world market that started at time t_0 . All regressions are estimated using the Cox Proportional Hazard Model (Cox, 1972) and control for exporting-country and time fixed effects. Estimations also allow for different baseline hazard rate across (4)-(6) for the subsample of countries that experienced stock market liberalization during the 1980-1997 period. Columns (1) and (4) report results for all countries with available data on the legal index "Judicial Efficiency". Columns (2) and (5) report results for countries with below-median value for this legal index. Columns (3) and (6) and not hazard ratios. Robust standard errors clustered at (exporting country)^{*}time $(c * t_0)$ level are in parentheses. *, **, and *** denote statistical significance at the

10%, $5%$, and $1%$ levels, respectively.						
	(1)	(2)	(3)	(4)	(5)	(9)
	All	All	All	Lib	Lib	Lib
	legal index	legal index	legal index	legal index	legal index	legal index
	available	below median	above median	available	below median	above median
StM liberalization x Distance to comparative advantage	0.082^{***}	0.050	0.191^{***}	0.069^{**}	0.042	0.097^{**}
	(0.026)	(0.030)	(0.051)	(0.029)	(0.042)	(0.040)
StM liberalization	-1.447***	-1.518^{***}	-1.317^{***}	-1.658^{***}	-1.568^{***}	-1.736^{***}
	(0.053)	(0.065)	(0.098)	(0.061)	(0.085)	(0.085)
Distance to comparative advantage	0.029	0.038^{*}	-0.059	0.036^{*}	0.026	0.068^{**}
	(0.018)	(0.020)	(0.043)	(0.019)	(0.028)	(0.027)
Full set of controls	Yes	Yes	Yes	$\mathbf{Y}_{\mathbf{es}}$	Yes	\mathbf{Yes}
Exporting country FE	Y_{es}	\mathbf{Yes}	Yes	Yes	${ m Yes}$	Yes
Time FE	Y_{es}	\mathbf{Yes}	Yes	Yes	\mathbf{Yes}	Yes
Product stratification	Yes	\mathbf{Yes}	Yes	Yes	\mathbf{Yes}	$\mathbf{Y}_{\mathbf{es}}$
Observations	38.005	27.553	10.452	32.992	15.119	17.873

M
La
o fo
tule
f B
n c
dia
Me
he
of t
ve (
bo
d A
an
MO
Bel
al,
viv
Sur
t,
xpo
É
ern
E E
luor
ЧI
an
ion
izat
rali
ibe
ît L
urke
M_{∂}
ock
Stc
XI:
le .
Tab

The dependent variable is the hazard rate for an export spell of product k from country c to the world market that started at time t_0 . All regressions are estimated using the Cox Proportional Hazard Model (Cox, 1972) and control for exporting-country and time fixed effects. Estimations also allow for different baseline hazard rate across products by defining product k as strata variable. Variables are defined in Table 3 in the main text. Columns (1)-(3) show results for the whole sample and columns (4)-(6) for the subsample of countries that experienced stock market liberalization during the 1980-1997 period. Columns (1) and (4) report results for all countries with available data on the legal index "Rule of Law". Columns (2) and (5) report results for countries with below-median value for this legal index. Columns (3) and (6) report results for countries with above-median value for this legal index. Full set of controls corresponds to column (4) of Table 3 in the main text. We report coefficients and not hazard ratios. Robust standard errors clustered at (exporting country)*time $(c * t_0)$ level are in parentheses. *, **, and *** denote statistical significance at the 10%,

5%, and 1% levels, respectively.						
	(1)	(2)	(3)	(4)	(5)	(9)
	All	All	All	Lib	Lib	Lib
	legal index					
	available	below median	above median	available	below median	above median
StM liberalization x Distance to comparative advantage	0.082^{***}	0.002	0.373^{***}	0.069^{**}	0.009	0.177^{***}
	(0.026)	(0.029)	(0.050)	(0.029)	(0.034)	(0.039)
StM liberalization	-1.447^{***}	-1.511^{***}	-1.525^{***}	-1.658^{***}	-1.709^{***}	-1.649^{***}
	(0.053)	(0.060)	(0.120)	(0.061)	(0.078)	(0.089)
Distance to comparative advantage	0.029	0.050^{***}	-0.258***	0.036^{*}	0.075^{***}	-0.058^{*}
	(0.018)	(0.019)	(0.051)	(0.019)	(0.022)	(0.035)
Full set of controls	Yes	Yes	Yes	Yes	Yes	$\mathbf{Y}_{\mathbf{es}}$
Exporting country FE	Yes	Yes	\mathbf{Yes}	\mathbf{Yes}	${ m Yes}$	Yes
Time FE	Yes	Yes	Yes	\mathbf{Yes}	\mathbf{Yes}	Yes
Product stratification	Yes	\mathbf{Yes}	Yes	Yes	\mathbf{Yes}	\mathbf{Yes}
Observations	38.005	30.238	7.767	32.992	20.835	12.157

	Ĩ	xpropriation				
The dependent variable is the hazard rate for an export spell of the Cox Proportional Hazard Model (Cox, 1972) and control fc products by defining product k as strata variable. Variables a	f product k from or exporting-con are defined in T	m country c to the untry and time fix Table 3 in the mai	world market that ed effects. Estimat n text. Columns	t started at tim tions also allow (1)-(3) show re	te t_0 . All regression for different baselin sults for the whole	s are estimated using ne hazard rate across sample and columns
(4)- (6) for the subsample of countries that experienced stock with available data on the legal index "Risk of Expropriation".	market liberali . Columns (2)	ization during the and (5) report res	1980-1997 period sults for countries	I. Columns (1) with below-mee	and (4) report res lian value for this l	ults for all countries egal index. Columns
(3) and (6) report results for countries with above-median val report coefficients and not hazard ratios. Robust standard erro	lue for this leg ors clustered at	al index. Full set (exporting count	of controls corres y^{*} time $(c * t_0)$ le	ponds to colun vel are in parer	in (4) of Table 3 ir itheses. *, **, and ^{>}	the main text. We *** denote statistical
significance at the 10%, 5%, and 1% levels, respectively.						
	(1)	(2)	(3)	(4)	(5)	(9)
	All	All	All	Lib	Lib	Lib
	legal index	legal index	legal index	legal index	legal index	legal index
	available	below median	above median	available	below median	above median
StM liberalization x Distance to comparative advantage	0.082^{***}	0.002	0.373^{***}	0.069^{**}	0.009	0.177^{***}
	(0.026)	(0.029)	(0.050)	(0.029)	(0.034)	(0.039)
StM liberalization	-1.447^{***}	-1.511^{***}	-1.525^{***}	-1.658^{***}	-1.709^{***}	-1.649^{***}
	(0.053)	(0.060)	(0.120)	(0.061)	(0.078)	(0.089)
Distance to comparative advantage	0.029	0.050^{***}	-0.258^{***}	0.036^{*}	0.075^{***}	-0.058*
	(0.018)	(0.019)	(0.051)	(0.019)	(0.022)	(0.035)
Full set of controls	Yes	\mathbf{Yes}	\mathbf{Yes}	Yes	$\mathbf{Y}_{\mathbf{es}}$	Yes
Exporting country FE	Y_{es}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	Y_{es}	Yes
Time FE	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	Yes
Product stratification	Yes	${ m Yes}$	\mathbf{Yes}	Yes	Yes	Yes
Observations	38,005	30,238	7,767	32,992	20,835	12,157

Table XII: Stock Market Liberalization and Long-Term Export Survival, Below and Above of the Median of Risk of

The dependent variable is the hazard rate for an export spell of the Cox Proportional Hazard Model (Cox, 1972) and control ft products by defining product k as strata variable. Variables a (4)-(6) for the subsample of countries that experienced stock with available data on the legal index "Risk of Repudiation". (3) and (6) report results for countries with above-median va- report coefficients and not hazard ratios. Robust standard erro significance at the 10%, 5%, and 1% levels, respectively.	f product k fron or exporting-co are defined in T market liberal Columns (2) a ulue for this leg ors clustered at	m country c to the untry and time fix fable 3 in the mai ization during the und (5) report resu al index. Full set (exporting countr	world market tha ed effects. Estima n text. Columns i 1980-1997 period ilts for countries v of controls corres $y)^*$ time $(c * t_0)$ le	t started at tim tions also allow (1)-(3) show ree (1) Columns (1) with below-med sponds to colur svel are in parer	the t_0 . All regressions for different baselin sults for the whole and (4) report res and (4) of Table 3 ir in (4) of Table 3 ir itheses. *, **, and *	s are estimated using the hazard rate across sample and columns ults for all countries egal index. Columns the main text. We *** denote statistical
	(1) All	(2) All	(3) All	(4) Lib	(5) Lib	(6) Lib
	legal index	legal index	legal index	legal index	legal index	legal index
	available	below median	above median	available	below median	above median
StM liberalization x Distance to comparative advantage	0.082^{***}	0.043	0.144^{***}	0.069^{**}	0.042	0.130^{***}
	(0.026)	(0.029)	(0.056)	(0.029)	(0.039)	(0.038)
StM liberalization	-1.447^{***}	-1.595^{***}	-1.156^{***}	-1.658^{***}	-1.696^{***}	-1.661^{***}
	(0.053)	(0.063)	(0.105)	(0.061)	(0.076)	(0.088)
Distance to comparative advantage	0.029	0.041^{**}	-0.045	0.036^{*}	0.029	-0.003
	(0.018)	(0.019)	(0.058)	(0.019)	(0.023)	(0.034)
Full set of controls	Yes	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$	Yes	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$
Exporting country FE	Yes	Yes	\mathbf{Yes}	\mathbf{Yes}	Y_{es}	Yes
Time FE	Yes	Yes	\mathbf{Yes}	\mathbf{Yes}	Yes	Yes
Product stratification	Yes	\mathbf{Yes}	\mathbf{Yes}	Yes	Yes	${ m Yes}$
Observations	38.005	30.838	7.167	32.992	21.845	11.147

Table XIII: Stock Market Liberalization and Long-Term Export Survival, Below and Above of the Median of Risk of Repudiation

15

OD	
ŝĊti	
ote	
$\mathbf{P}_{\mathbf{r}}$	
or	
este	
ПV€	
Ļ	
u o	
liaı	-
Iec	
2	
$^{\mathrm{th}}$	
of	
Ve	
poq	
A l	5
und	
N N	
ole	:
Ã	
al,	
viv	
Jur	
+2	
10C	
EX	
n]	
en	
<u> </u>	
on	•
Γ	:
anc	
n å	
tic	
iza	
ral	•
ibe	
Ţ	
ket	
lar	
N	
ock	2
$\mathbf{\hat{s}}$	
$\stackrel{\sim}{>}$	
XI	
ole	
Lab	
	Ē

The dependent variable is the hazard rate for an export spell	of product k from	country c to the w	orld market that start	ed at time t_0 .	All regressions are estimat	ted using
the Cox Proportional Hazard Model (Cox, 1972) and control	for exporting-cour	itry and time fixed	effects. Estimations a	lso allow for di	fferent baseline hazard ra	te across
products by defining product k as strata variable. Variables	are defined in Ta	ble 3 in the main	text. Columns (1)-(3)	show results f	or the whole sample and	columns
(4)- (6) for the subsample of countries that experienced stock	market liberalizat	ion during the 198	D-1997 period. Column	s (1) and (4) r	eport results for all count	ries with
available data on the legal index "Investor Protection". Colum	nns (2) and (5) rep	ort results for com	tries with below-medi	un value for thi	s legal index. Columns (3)) and (6)
report results for countries with above-median value for this l	egal index. Full se	t of controls corres	ponds to column (4) o	f Table 3 in the	i main text. We report co	efficients
and not hazard ratios. Robust standard errors clustered at (ϵ	exporting country)	*time $(c * t_0)$ level	are in parentheses. *,	**, and *** de	mote statistical significan	ce at the
10%, 5%, and 1% levels, respectively.						
	11	(0)	(0)		11	

10%, 5%, and 1% levels, respectively.						
	(1)	(2)	(3)	(4)	(5)	(9)
	All	All	All	Lib	Lib	Lib
	legal index	legal index	legal index	legal index	legal index	legal index
	available	below median	above median	available	below median	above median
StM liberalization x Distance to comparative advantage	0.082^{***}	0.034	0.209^{***}	0.069^{**}	0.039	0.127^{***}
	(0.026)	(0.029)	(0.057)	(0.029)	(0.041)	(0.039)
StM liberalization	-1.447^{***}	-1.563^{***}	-1.268^{***}	-1.658^{***}	-1.670^{***}	-1.687^{***}
	(0.053)	(0.062)	(0.110)	(0.061)	(0.078)	(0.088)
Distance to comparative advantage	0.029	0.041^{**}	-0.131^{**}	0.036^{*}	0.036	-0.064^{*}
	(0.018)	(0.018)	(0.056)	(0.019)	(0.023)	(0.036)
Full set of controls	Yes	Yes	$\mathbf{Y}_{\mathbf{es}}$	Yes	Yes	$\mathbf{Y}_{\mathbf{es}}$
Exporting country FE	\mathbf{Yes}	\mathbf{Yes}	Yes	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}
Time FE	Yes	Yes	Y_{es}	\mathbf{Yes}	Yes	Yes
Product stratification	Yes	\mathbf{Yes}	\mathbf{Yes}	Yes	Yes	${ m Yes}$
Observations	38,005	31,216	6,789	32,992	21,197	11,795